

Directorate General NDRF & Civil Defence (Fire) Ministry of Home Affairs East Block 7, Level 7, NEW DELHI, 110066

Fire Hazard and Risk Analysis in the Country for Revamping the Fire Services in the Country

Final Report - User/ Training Manual – Fire Decision Support System (FDSS)

November 2012

Submitted by

RMSI A-8, Sector 16 Noida 201301, INDIA Tel: +91-120-251-1102, 2101 Fax: +91-120-251-1109, 0963 <u>www.rmsi.com</u>

Contact: **Sushil Gupta** General Manager, Risk Modeling and Insurance Email: Sushil.Gupta@rmsi.com





Table of Contents

Та	ble of Co	ntents	2
Lis	t of Figu	es	5
Lis	t of Table	es	9
1	Overvi	ew	10
	1.1 Ab	out this Manual	10
2	Getting	Started	12
	2.1 Lo	gging into FDSS	12
	2.1.1	Normal User login	13
	2.1.2	Administrator Login	14
	2.1.3	Changing user-account information	14
	2.1.4	Logging Out	14
3	Introdu	cing the FDSS interface	15
	3.1 To	olbar panel	16
	3.1.1	Generate Maps tool	18
	3.1.2	Generate Fire Statistical Report	23
	3.1.3	National Infrastructure Report tool	26
	3.1.4	National Gap Report tool	27
	3.1.5	Help tool	28
	3.1.6	User Management tool	28
	3.1.7	Floating Navigation tool pane	28
:	3.2 Ma	ap window	30
:	3.3 Ma	ap Tools window	30
	3.3.1	How to navigate the tree view	31
	3.3.2	Layers	32
	3.3.3	Legends overview	37
:	3.4 Se	arch window	38
	3.4.1	Search	39
	3.4.2	Adjacent Fire Station Search	44
	3.4.3	Legends	46
	3.4.4	Select by Attributes	47
	3.4.5	Select by Location	52
	3.5 Ar	alysis window	58
	3.5.1	Infrastructure	58
	3.5.2	Gap Analysis	61



	3.6	Res	ults window	64
	3.7	Мар	Elements	66
4	Viev	wing	Hazard Zones, and Fire Risk	68
	4.1	Viev	ving Hazard Zones at the Country Level	68
	4.2	Viev	ving District Fire Risk at the Country-Level	69
5	Lay	ers		70
	5.1	Haz	ard Layers	70
	5.1.	1	Wind Zones	70
	5.1.	2	Seismic Zones	71
	5.1.	3	Climatic Zones	71
	5.2	Cou	intry layer	72
	5.2.	1	Fire Station Density in India	72
	5.2.	2	Operational Fire Stations	73
	5.2.	3	Proposed Fire Stations	74
	5.2.	4	Other Agency Fire Stations Area	75
	5.2.	5	District Fire Risk Category	76
	5.3	Stat	es and Union Territories layer	77
	5.3.	1	Major Roads	78
	5.3.	2	Minor Roads	79
	5.3.	3	Rail Network	80
	5.3.	4	Land Use Land Cover	81
	5.3.	5	Built-up Area	82
	5.3.	6	Ideal Urban Fire Station Jurisdiction	83
	5.4	Adn	ninistrative Boundaries layer	84
	5.4.	1	Phase-Wise States	84
	5.4.	2	State/UT Capital	85
	5.4.	3	Districts	86
6	Use	r Ma	nagement	87
	6.1	Cha	inge password	88
	6.2	Cha	inge Group	89
	6.3	Add	New User	89
	6.4	Ren	nove an existing user	90
7	Ann	exur	e: FDSS desktop application	92
	7.1	Data	a Import Utility tool	92
	7.1.	1	Opening the tool	92
	7.1.	2	Data Import Utility tool interface	93
	7.2	Mas	ter Table Editing Tool1	104



7.2.1	Opening the tool)4
7.2.2	Master Table Editing Tool interface10)5



List of Figures

Figure 2-1 : Launching the FDSS application in web browser (Login Page)	12
Figure 2-2: FDSS Home Page	13
Figure 3-1: FDSS User Interface showing its main parts	15
Figure 3-2: FDSS Toolbar panel	16
Figure 3-3: Generate Maps tool pop-up window displaying layers	19
Figure 3-4: Selecting State, District and Layers in the Generate Maps tool	20
Figure 3-5: Sample output of Generate Map tool	21
Figure 3-6: Opening or saving the generated report in Excel or PDF format	22
Figure 3-7: Choose Helper Application dialog box	23
Figure 3-8: Generate Fire Statistical Report tool pop-up window displaying layer inform	
Figure 3-9: Selecting State, District, Month and Year in the Generate Fire Statistical F tool	Report
Figure 3-10: Sample output of Generate Fire Statistical Report tool	25
Figure 3-11: Opening or saving the generated report in PDF format	26
Figure 3-12: Opening or saving the generated report in Excel format	26
Figure 3-13: First page of National Infrastructure Report showing report export option	27
Figure 3-14: First page of National Gap Report showing report export option	28
Figure 3-15: Floating Navigation toolbar	29
Figure 3-16: Map Tools window with layers and legends	31
Figure 3-17: Navigating the tree view	31
Figure 3-18: Default tree view of layers in Map Tools window	32
Figure 3-19: Base Layers when connected to internet	33
Figure 3-20: FDSS - Google (Satellite) Map	33
Figure 3-21: Base Layers when not connected to internet	34
Figure 3-22: FDSS- India Map	34
Figure 3-23: Fully expanded tree view of Hazard layers	35
Figure 3-24: Fully expanded tree view of Country layer	35
Figure 3-25: Expanded tree view of States and Union Territories	36
Figure 3-26: Expanded tree view of the Administrative Boundaries layer	37
Figure 3-27: Expanded Legends tab in Map Tools window showing Seeker bar	38
Figure 3-28: Search window displaying search options	39
Figure 3-29: Fire Station Search functionality	40
Figure 3-30: Zoom into a selected district	41
Figure 3-31: Search results table showing Map It, Clear and Report options	42



Figure 3-32: Fire Station search showing populated table and corresponding result mapped in the Map window after using the Map It button
Figure 3-33: Sample of a searched fire station report
Figure 3-34: Adjacent Fire Station Search window
Figure 3-35: Sample report of Adjacent Fire Station Search tool
Figure 3-36: Legends tab in Search window illustrating legend of wind zones
Figure 3-37: User Interface of Select by Attribute tool
Figure 3-38: Layer information generated in Map dropdown
Figure 3-39: Example showing use of Select by Attribute tool
Figure 3-40: Map showing area lying under Zone II after unchecking the checkbox corresponding to Seismic Zones layer
Figure 3-41 : User Interface for Select by Location tab
Figure 3-42: Activating the layer
Figure 3-43: Filling layer information
Figure 3-44 : Map showing operational fire stations that are within a distance of 10 km from proposed fire stations
Figure 3-45: Analysis window showing analysis tools
Figure 3-46: Analysis window illustrating State and District selections
Figure 3-47: Sample page from Infrastructure report for Delhi State (All Districts selected). 60
Figure 3-48: Sample page from Infrastructure report of Central district, Delhi state (Only Central district selected)
Figure 3-49: Sample page from Gap Analysis report for Delhi state (All districts selected) 63
Figure 3-50: Sample page from Gap Analysis report of Central district, Delhi state (Only Central district selected)
Figure 3-51: Results window showing the populated results with the use of Identify a feature tool
Figure 3-52: Map elements in FDSS User Interface
Figure 4-1: Viewing levels of seismic hazard risk (Seismic zones) at country level
Figure 4-2: Map showing district fire risk category in India
Figure 5-1: Tree Structure of the Hazard Layers70
Figure 5-2: Map showing wind zones in India70
Figure 5-3: Map showing seismic zones in India71
Figure 5-4: Map showing climatic zones in India
Figure 5-5: Fully expanded tree view of the Country layer
Figure 5-6: Fire Station Density in India
Figure 5-7: Operational Fire Stations in India
Figure 5-8: Proposed Fire Stations in India
Figure 5-9: Other Agency Fire Stations Area
Figure 5-10: District Fire Risk Category in India77



Figure 5-11: Expanded view of Delhi state illustrating sub-layers	78
Figure 5-12: Major Roads of Delhi	78
Figure 5-13: Minor Roads of Delhi	79
Figure 5-14: Rail Network of Delhi	80
Figure 5-15: Land Use Land Cover of Delhi	81
Figure 5-16: Built-up area of Delhi	82
Figure 5-17: Ideal Urban Fire Station Jurisdictions of Delhi	83
Figure 5-18: Expanded Tree view of Administrative Boundaries layer	84
Figure 5-19: Map showing phase wise states in India	84
Figure 5-20: Map showing State Capital	85
Figure 5-21: Map showing districts in India	86
Figure 6-1: User Management window	87
Figure 6-2: User Management window illustrating Administrator tasks	88
Figure 6-3: Change Password window	88
Figure 6-4: Change Group window	89
Figure 6-5: Create New User window	90
Figure 6-6: Remove user prompt window	90
Figure 7-1: Data Import Utility tool UI	93
Figure 7-2: Window highlighting Data Import Utility tool interface	94
Figure 7-3: File menu with its sub menu	94
Figure 7-4: Window showing File menu to add new fire station (highlighted by red rectain	•
Figure 7-5: Form showing Fire Station General Information page	96
Figure 7-6: Form showing Fire Calls and Incidence Details page	97
Figure 7-7: Message box displayed on successful addition of a fire station	97
Figure 7-8: Dialog box showing question to update ideal fire station details	98
Figure 7-9: Window showing list of all the proposed fire stations within 30 km radius o new fire station	
Figure 7-10: Window showing existing fire station details of Delhi state	. 100
Figure 7-11: Window showing selection of a row corresponding to a fire station having ic with context menu (highlighted by red rectangle)	
Figure 7-12: Form showing general information of Shashtri Park fire station, Delhi	. 102
Figure 7-13: Form showing Fire Calls and Incidence Details page with update be (highlighted by red rectangle)	
Figure 7-14: Message box showing message to update fire station details	. 103
Figure 7-15: Dialog box prompting user to confirm or ignore deletion of selected fire st	
Figure 7-16: Master Table editing tool UI	. 104



Figure 7-17: Window showing list of all ideal fire stations in India	105
Figure 7-18: Window showing details of ideal fire stations in Pune, Maharashtra	106
Figure 7-19: Window showing selection of a row corresponding to fire station id 613	107
Figure 7-20: Window showing selection of Urban/Rural column to be edited	108
Figure 7-21: Window showing Urban/Rural column changed from Urban to Rural type for station having Id 613	
Figure 7-22: Window showing selection of a row corresponding to fire station id 591	110
Figure 7-23: Dialog box displaying question to delete a fire station	110



List of Tables

Table 3-1: Tools in FDSS Toolbar panel and their functions	16
Table 3-2: Various tools and their functions in Floating Navigation tool pane	29
Table 3-3: Details of conditions in the Select by Attributes tool	51
Table 3-4: Details of spatial functions in Select by Location tool	56



1 Overview

Fire service is one of the most important emergency response services. It is the 'first responder' to any kind of emergency. The growth of this emergency service in the country has been on an ad-hoc basis, without much scientific analysis of existing risks in different parts of the country. Now, varying risk scenarios need different kinds and types of equipment depending upon the risk category of the coverage area of the Fire Station. Lack of knowledge management for future planning, institutional capacity and funds are major challenges in addressing improvements in fire and emergency services in the country.

As per a recent analysis by the Standing Fire and Advisory Council (SFAC), the overall deficiency in the country is more than 80% in fire fighting and rescue vehicles and 95% in fire stations and fire personnel, respectively (NDMA Guideline, 2012, CR SFAC, 2011).

The Directorate of National Disaster Response and Civil Defence (NDRF&CD, Fire Cell), Ministry of Home Affairs (MHA) felt the need for a comprehensive study to identify existing gaps considering the above said deficiency, rapid pace of urbanization and industrialization in the country and the increasing fire risks from various hazards such as Fire Following an Earthquake (FFEQ).These gaps can be in terms of availability and requirement of fire stations, capacity-building, not only in terms of trained man-power but also vehicles, specialized equipment for fire-fighting and rescue. This comprehensive study aims to prepare a perspective plan for the next 10 years for revamping the fire services in the country.

The **Fire Decision Support System** (FDSS) has been developed mainly as a dynamic application, aimed at supporting decision makers take optimal decisions on complex tasks, such as resource prepositioning, gap analysis, prioritization, and resource optimization along with the day-to-day tasks. The most important aspect of FDSS is that it enables the apex fire management authority to provide the entire country's fire agencies information on a single platform.

1.1 About this Manual

The FDSS User Manual aims to assist users in using the system effectively.

There are six sections in the manual which are arranged as follows: Section 1 (Overview) provides an overview of FDSS. Section 2 (Getting Started) is devoted to take the user through the processes of logging into the system and logging out of it. Section 3 (Introducing the FDSS Interface) introduces the various displays and controls used in the User Interface. Section 4 (Viewing Hazard and Climatic Zones, and Fire Risk) takes the user through the basic steps of using FDSS to view the wind and seismic hazard zones, and the climatic zones at a country level, and for viewing the fire risk at a district level. The section also details steps for generating reports thereof. Section 5 describes all the layers in FDSS, namely, Hazard, Country, States and Union Territories, and Administrative Boundaries etc. These may actually be looked upon as an extension of the FDSS interface section (Section 3) where these layers are introduced summarily only. Section 6 deals with the User Management functions provided in FDSS and are primarily of use to system administrators.



Hence, Sections 2 to 4 and Section 6 pertain to assisting users with the various tasks they can perform through FDSS. These sections also serve as a reference to any queries related to using the displays and controls provided in the User Interface. The instructions in these sections are in the form of well defined and easy to follow steps. Section 5 forms the descriptive part of the manual, wherein all the layers in FDSS application are described to give users an idea of what each layer includes.



2 Getting Started

2.1 Logging into FDSS

Open the web browser¹ and type the URL in the address bar to launch the FDSS application as shown by the red highlighted box in Figure 2-1.The URL depends upon whether the application is accessed across the internet or installed locally. The FDSS administrator provides the URL for all new users.

🗲 🛞 pieronline :msi.com /FDSS/Login.aspx	⊽ C Soogle	۶ 🍂
🌇 Fire Decision Support System		
Login		
Username: Password:		
Group: Administrator		
Administration Normal User pi		
A product of		
A product of RMSI Similar Copyright @ RMSI Private Limited, India		

Figure 2-1 : Launching the FDSS application in web browser (Login Page)

The user is taken to the FDSS login page as shown in Figure 2-1. The login page has three fields namely, the Username textbox, the Password textbox and the Group dropdown list. FDSS has two user groups; Administrator and Normal User. The Group dropdown displays Administrator by default. The user can click the down arrow \checkmark of the dropdown to reveal the list displaying the second user group option, namely Normal User.

¹ Firefox 3.0 or above is recommended, though, IE 7.0 or above is compatible



2.1.1 Normal User login

New users can only login to FDSS after the administrator has created user accounts giving unique user names and passwords. The user can then use these credentials and login to FDSS by following the steps described next.

Type user name in the Username textbox, the password in the Password textbox, provided by the administrator, and select the Normal User option from the Group dropdown.

Click the Login button. The application validates the user name and password provided. If these are correct, the user is taken to the FDSS home page as shown in Figure 2-2. The logged in user is identified by the user name at the top right of the screen as: You are logged in as:
user name>



Figure 2-2: FDSS Home Page



2.1.2 Administrator Login

Type in the administrator user name in the Username textbox, the password in the Password textbox, and select the Administrator option from the Group dropdown. FDSS is shipped with a number of users already created, when the system is first installed or configured. The illustration in Figure 2-1 shows the user logging in as an administrative user.

Click the Login button. The application validates the user name and password provided. If these are correct, the user is taken to the FDSS main page as shown in Figure 2-2. The logged in user is identified by the user name at the top right of the screen as: You are logged in as:
user name>

2.1.3 Changing user-account information

A normal user cannot change the Username and Password. Only a user from the Administrator group can modify user account information. The user can continue to use the login information as configured by the administrator or request the administrator to change such account information. Please refer to Section 6 for more information.

2.1.4 Logging Out

To log out of the FDSS application, click on the [Log Out] link at the top right corner of the screen as shown in Figure 2-2.



3 Introducing the FDSS interface

The FDSS User Interface has seven parts as listed below:

- 1. Toolbar panel
- 2. Map window
- 3. Map Tools window
- 4. Search window
- 5. Analysis window
- 6. Results window
- 7. a) Scale bar
 - b) Map Co-ordinates

When a user logs into a new FDSS session, results window pane is kept hidden. User can view it by clicking the expand icon in the bottom right corner of the page. All other windows are arranged in the User Interface as shown in Figure 3-1.



Figure 3-1: FDSS User Interface showing its main parts



3.1 Toolbar panel

The toolbar panel shown in Figure 3-2 contains a set of tools for working with maps and their content. These tools help in navigating the Map window (Section 3.2). The panel also includes tools to generate user defined maps, National Level Infrastructure report and National Level Gap report. The tool name and brief description is displayed when a user points mouse over a tool icon. These tools are summarily described in Table 3-1.



Figure 3-2: FDSS Toolbar panel

Tool Icon	Tool Name	Description
	Zoom to India	This tool displays the map of India in the Map window
	Zoom In	Zoom in to the map, increasing the scale and magnification. User can click the left mouse button to zoom or user can drag the mouse to create and zoom into a rectangular area.
. V	Zoom Out	Zoom out of the map. User can click the left mouse button to zoom out or user can drag the mouse to create and zoom out of a rectangular area.
	Pan Map	This tool helps in panning the Map for navigating to different locations. At start up, control rests with the Pan tool by default.
	Previous View	This tool takes the user to the previous view in the Map window.

Table 3-1: Tools in FDSS Toolbar panel and their functions.



Tool Icon	Tool Name	Description
	Next View	This tool takes the user to the next view in the Map window.
	Measure Distance	This tool helps in measuring distance on the map. User can mark the points and double click to measure the distance.
	Measure Area	This tool helps in measuring area on the map. User can mark the points and double click to measure the area.
	Measure by Circle	This tool helps in measuring the area of any region in a user defined circle. Marking the radius and double clicking calculates the area of the circular region.
	Identify a feature	This tool helps in displaying information related to a geographical location in the Map window and the selected layer in the Map Tools window. See Section 3.6 for usage of this tool.
	Generate Map	This tool generates map as specified by the user. This tool is described in Section 3.1.1 below.
	Fire Statistical Report	This tool generates Fire Statistical Report as specified by user. This tool is described in Section 3.1.2 below.



Tool Icon	Tool Name	Description
	National Infrastructure Report	This tool generates National level infrastructure report of India. This tool is described in Section 3.1.3 below.
	National Gap Report	This tool generates National level gap analysis report of India. This tool is described in Section 3.1.4 below.
2	Help	This tool generate user manual for providing help to use FDSS. This tool is described in Section 3.1.4 below.
8	User	This tool helps in basic user management tasks. This tool is described in Section 3.1.6 below.

3.1.1 Generate Maps tool

The Generate Maps tool in the Toolbar panel is used to prepare thematic maps as defined by the user. To generate a map follow the steps mentioned below:

1. Click on the Generate Maps tool from the toolbar to display the Generate Maps window. A list of layers available in FDSS is displayed as shown in Figure 3-3. The Hazard and Administrative Boundaries layers are displayed in an expanded view by default. User can expand the Country, States, and Union Territories layers by clicking

the expand icon \blacksquare to the right of the layer name. See Section 3.3.1 for assistance with operating the tree structure.



Generate	Maps		×
State:	Select State	~	
District:	Select District	~	
	ers azard Layers Wind Zones Seismic Zones Climatic Zones ountry tates and Union Territories dministrative Boundaries Phase-Wise States State Capital Districts		
	Ge	nerate Map	Cancel

Figure 3-3: Generate Maps tool pop-up window displaying layers

2. Select state from the State dropdown by clicking the dropdown icon. User can either select a single state or the All option from the State dropdown to generate a map at country level. Similarly, user has to select a district from the District dropdown list when a particular state is selected. As an example, if the user wishes to generate a map of wind zones in Andhra Pradesh for all the districts then the user has to select Andhra Pradesh state from the State dropdown and the All option from the District dropdown and then click the checkbox corresponding to Wind Zones layer as shown in Figure 3-4. Finally, the user has to click the Generate Map button to view the final thematic map.

Generate	Maps			X
State:	Andhra Pradesh		~	
District:	A		~	
	azard Layers V Wind Zones Seismic Zones Climatic Zones Duntry tates and Union Terri dministrative Bounda Phase-Wise Sta State Capital Districts	ries		
		Generate Ma	ap	Cancel

Figure 3-4: Selecting State, District and Layers in the Generate Maps tool

3. A map of checked layer(s) for the state and district specified is generated in a separate browser page. A sample map of wind zones in Andhra Pradesh state for All districts is shown in Figure 3-5.





Figure 3-5: Sample output of Generate Map tool

3.1.1.1 Exporting and Printing a Generated Map or Report

As highlighted by the red box in Figure 3-5, the user can save the generated map in Acrobat (PDF) file format by clicking the Export button as highlighted by the blue circle. In FDSS application, the user may find a number of instances where similar options for generating and exporting reports are provided. The following steps are useful in exporting and printing such maps and reports.

Select the Excel² or Acrobat (PDF) file type from the Export Select a format Export dropdown list at the top of the Report window as highlighted by the red box in Figure 3-5. Selecting any one of the options enables the Export button to the right of the dropdown list.

² Excel option will be available only in case of reports. For maps the available export option is Acrobat (PDF) only.



2. Click the Export button. This displays a new dialog box (Opening...) as shown in Figure 3-6.

Opening GenMap.	pdf	×
You have chosen to c	ipen:	
	obe Acrobat 7.0 Document eronline.rmsi.com	
What should Firefox	do with this file?	
O Open with	Adobe Reader (default)	
⊙ <u>S</u> ave File	Adobe Reader (default)	
📃 Do this <u>a</u> uto	Other	
	OK Cancel	

Figure 3-6: Opening or saving the generated report in Excel or PDF format

- 3. Select the Open with option button. By default, the Microsoft Office Excel (default) or Adobe Reader (PDF) option is selected in the file type dropdown list. See the footnotes³ for more information.
- 4. Click the OK button to open the report as an Excel or PDF document.
- 5. Select the Save As or Save File option button. By default, the Microsoft Office Excel (default) or Adobe Reader (PDF) option is selected in the file type dropdown list. See the notes under this section for more information.
- 6. Click the OK button to save the report as an Excel or PDF document. The user can then use the Save and Print options to save and print the file as for other Windows based application files.

³ The default value in the file type dropdown (Figure 3-6) list corresponds to the selection made by the user in Step 1 above.

The file type dropdown list for the Open with option (Figure 3-6) also has the "Other" option. Selecting this option displays the Choose Helper Application dialog box (Figure 3-7). A sample of this dialog box is shown below. The user can use this dialog box to select an appropriate application to open/save the report. HOWEVER, THE USE OF THIS OPTION IS NOT RECOMMENDED.



Choose	Helper Application
POF	Adobe Acrobat 7.0 Document GenMap.pdf
Send th	his item to:
٨	Adobe Reader
Ø	Internet Explorer
1	Firefox
	ArcGlobe Application
	ArcMap 💌
Brow	ise OK Cancel

Figure 3-7: Choose Helper Application dialog box

3.1.2 Generate Fire Statistical Report

in the Toolbar panel is used to prepare Fire The Generate Fire Statistical Report tool Statistical Report as defined by the user. To generate a Report follow the steps mentioned below:

1. Click on the Generate Fire Statistical Report tool from the toolbar to display the Generate Fire Statistical Report window. A list of layers available in FDSS is displayed as shown in Figure 3-8. The State, District, Month, and Year layers are displayed in a view by default.

State:	Select State	~	
District:	Select District	~	
Month:	Select Month	•	
Year:	Select Year	~	

Figure 3-8: Generate Fire Statistical Report tool pop-up window displaying layer information







2. Select state from the State dropdown by clicking the dropdown icon . User can either select a single state or the All option from the State dropdown to generate a Fire Statistical Report at country level. Similarly, user has to select a district from the District dropdown list when a particular state is selected. As an example, if the user wishes to generate a Fire Statistical Report of Andhra Pradesh State for all the districts for all months of Year 2011 then the user has to select Andhra Pradesh from the State dropdown, the All option from the District dropdown, the All option from the District dropdown as shown in Figure 3-9. Finally, the user has to click the Summary of Fire Statistics button to view the final Fire Statistical Report.

	ormation		
State:	Andhra Pradesh	~	
District:	All	~	
Month:	All	~	
Year:	2011	~	

Figure 3-9: Selecting State, District, Month and Year in the Generate Fire Statistical Report tool

3. A Report of selected layer(s) for the state, district, month and specified year is generated in a separate browser page. A sample Report of Andhra Pradesh state for All districts for all months in Year 2011 is shown in Figure 3-10.



						Excel Acrobat	(PDF) file							Di		te General y of Home A		
FDSS									N									
g Cl																		
					S	tate/UT L	evel l	Fire Sta	atistics F	Report								
State/UT	: And	nra Pradesh	1															
Number of Districts	: 23																	
Number of Fire Stations	: 251																	
Fire Statistics for 2011																		
Diatricta	Fire stations	Total Calls (A+B+ C+D)	Total Fire Catts (A)	Occupan	ry white brea	ik up of File Incide	nce Calla	Total Rescue Calls (8)	Break up	p of Reacue II	scidence C	alla	Special service calls (C)	Fatsel maticious calls (D)	Tels	i Injured	Tola	Deaths
			1007	Resid- ontial	Indus- trial	institutions? commercial	Others	Calling (C)	Road Accidents	Huilding Collepte	Animal	others	(C)		People Injured	Pire Personnel	People Diet	Pise Person
Adilabad	10	602	583	163	61	39	320	18		-		13				Injured		Del Divid
Anantapur	10	791	739	103	401	43	119	52	52			13	1			-		
Chittoor	15	791	159	170	401	40	110						-					
East Godavari	14	1,085	1.054	500		43	510	30			8	22	-		-		-	
Guntur	13	580	545	284	214	18	49	15	15							2		1
Hyderabad	10	897	840	209	85	244	302	57	57		-	2	2	2				
Karimnagar	8	508	497	107	8	18	384	9				9	-					
Khammam	8	323	319	129	0	11	173	4	2		1	1					-	1
Krishna	21	1,772	1,689	773	9	44	863	06	1		22	57	2	1		28		
Kurnool	12	1,241	1,181	322	635	62	162	60	60	-			-			2		
Mahabubnagar	8	638	625	92	9	23	501	11	8			3	2					
Medak	8	1,072	1.059	177	37	32	813	12	-		1	11	1		-		-	
Nalgonda	10	898	840	121	34	29	656	58	58				-					
Nizamabad	6	838	809	354	50	24	381	29	10	-	4	15	-			2	1	
Prakasam	11	-	-	-			+	-	-	-					•	-		6
Rangareddy	9	737	698	135	82	80	401	39	26		13		-		•	•		1
Sri Potti Sriramulu Nellore	12	203	202	43	98	7	54	1	5				-	•	•			
Srikakulam	12	550	544	242	7	23	272	6	-	-	•	6	2		•	5		1
Visakhapatnam	11	846	817	389	15	95	318	29	1		1	27	1	Ξ.		10		1
Vizianagaram	9	508	504	229	14	28	233	2	-	•	-	2	-	-		-		
Warangal	7	782	760	198	28	10	528	22	-	-	-	22	-				1	
West Godavari	13	1,212	1,181	512	6	48	615	31	2		5	24	-	÷		4	2	1
Y.S.R	12	-	84	-	-		-	-	2		14	1		-	-	-		
Total	251	16,059	15,486	5,133	1,800	921	7,632	565	297	+	55	212	6	2		59	3	9

Figure 3-10: Sample output of Generate Fire Statistical Report tool

3.1.2.1 Exporting a Generated Fire Statistical Report

As highlighted by the red box in Figure 3-10, the user can open or save the generated report in Acrobat (PDF) file format or Microsoft Excel file format by clicking the Export button as highlighted by the blue circle. In FDSS application, the user may find a number of instances where similar options for generating and exporting reports are provided. The following steps are useful in exporting such maps and reports.

- 1. Select the Excel or Acrobat (PDF) file type from the Export Select a format Select a format
- 2. Click the Export button. This displays a new dialog box (Opening...) as shown in Figure 3-11 and Figure 3-12.
- 3. Select the Open or Save button as shown in Figure 3-11 and Figure 3-12 to open or save the Fire Statistical Report.



File Dow	rnload 🛛 🔀
Do you	want to open or save this file?
Addee	Name: StatisticalReport.pdf Type: Adobe Acrobat 7.0 Document From: pieronline.rmsi.com Open Save Cancel
2	While files from the Internet can be useful, some files can potentially harm your computer. If you do not trust the source, do not open or save this file, <u>What's the risk?</u>

Figure 3-11: Opening or saving the generated report in PDF format

File Dow	mload 🛛 🔀
Do you	want to open or save this file?
×	Name: StatisticalReport.xls Type: Microsoft Office Excel 97-2003 Worksheet From: pieronline.rmsi.com
	Open Save Cancel
?	While files from the Internet can be useful, some files can potentially harm your computer. If you do not trust the source, do not open or save this file. <u>What's the risk?</u>

Figure 3-12: Opening or saving the generated report in Excel format

3.1.3 National Infrastructure Report tool

The National Infrastructure Report tool in the toolbar generates an infrastructure report at the national level. The report opens in a separate browser page. The National Infrastructure Report is a comprehensive listing of details pertaining to all fire stations in India. The first page has demography statistics of India as shown in Figure 3-13. It is followed by state-wise infrastructure information of operational fire stations such as fire fighting and rescue vehicles, specialized equipment, available manpower by designation, and building infrastructure details.



The user can export the report in either Excel or Acrobat (PDF) file format as shown by the red highlighted box in Figure 3-13. See Section 3.1.1.1 for more detailed steps.

4 1 of 14 ▶ ▶ Select	a format 🕞 Export	
Select Excel	a format	Directorate General NDRF & CD (Fi
	t (PDF) file	Ministry of Home Affairs, Govt. of In
FDSS		6
- 1000		
	Infra	terusture Depart for India
	5 Shiril	structure Report for India
Country Demography		
Country	: India	
Number of States	: 35	
Number of Districts	: 640	
Number of Towns	: 7,935	
Number of Villages	: 640,867	
Total Number of Fire Stations	: 2,987 *	
Total Population (2011)	: 1,210,193,422	
Male Population (2011)	: 623,724,248	
Female Population (2011)	: 586,469,174	
Sex Ratio (Females per 1000 males)	: 940	
Population Density	: 382	
		ons have been relocated adjacent to currently operating Fire Stations
s - nazaru anu rusk ahalysis in ine country	with an objective to Prepare Capital live	stment and Institutional Strengthening Plan for revamping the Fire Services in the Country 1 of 14

Figure 3-13: First page of National Infrastructure Report showing report export option.

3.1.4 National Gap Report tool

and the

24.6
The National Gap Report Limit tool in the toolbar generates a Gap Analysis report at the
national level. The report opens in a new browser page. The National Gap Report is an
analytical summary of existing gaps in fire fighting and rescue vehicles, specialized
equipment, trained manpower, and building infrastructure. The report contains a state-wise
summation of gaps detailing additional requirement for operational, new urban, and new
rural fire stations. The report also contains financial details for filling such gaps and a 10-
year investment plan for entire India. A sample page of this report is shown in Figure 3-14.

User can export the report in either Excel or Acrobat (PDF) file format as indicated by red highlighted box in Figure 3-14. See Section 3.1.1.1(Exporting and Printing a Generated Map or Report) for help on exporting a report.



4 4 1	of 78 🕨	Sel Exc	ect a form ect a form el obat (PDF)	et	kpar	٩								D			NDRF & CD (F Affairs, Govt. of
🕺 FDS	S																6
						Gap	Analy	sis R	eport	for In	Idia						
						Contrast No. 14			ALC: MANAGE								
ehicle Detai	ls																
Current Vehic	le Invent	ory for Operatio	mal Fire S	Stations													
State	Fire Stations	Ideally Served Population Estimates	Water Tenders	Water Bowsers	Foam Tenders	Advanced Rescue Responders	Sky Lifts /TTL	DCP Tenders	Hose Tenders	BA Vans	Hazmat Vans	QRT	Motor Cycle Mists	Fire Boats	Ambu-lances	Education Vans	Total Vehicle
Andaman and Nicobar Islands	20	0	27	3	1	0	0	0	1	0	0	0	0	0	0	0	32
Andhra Pradesh	251	35,670,178	258	2	1	3	2	3	0	0	1	1	4	0	2	1	278
Arunachal Pradesh	13	416,547	26	2	1	0	0	0	0	0	0	1	0	0	0	0	30
Assam	110	18,371,991	268	0	4	1	1	0	0	0	2	3	0	0	0	0	279
Bihar	102	12,140,981	230	10	10	4	2	1	0	0	0	6	0	0	1	0	264
Chandigarh	7	753,890	10	4	0	3	2	3	0	0	0	5	7	0	3	0	37
Chhattisgarh	33	3,636,523	47	4	6	0	0	1	0	0	0	1	0	0	0	0	55
Dadra and Nagar Haveli	1	26,355	3	2	1	2	0	0	0	0	0	0	0	0	1	0	
Daman and Diu	3	129,888	4	1	2	2	0	0	0	0	0	1	1	0	3	0	1
Delhi	53	8,126,612	79	48	6	7	6	4	5	4	3	7	10	0	0	0	17
Goa	15	656,232	22	ा	2	3	1	0	0	0	0	2	0	0	10	0	4
Gujarat	183	26,096,729	300	241	29	36	11	5	0	1	0	5	0	9	134	0	77
Haryana	82	6,663,340	129	37	27	10	2	0	0	0	0	9	0	0	0	0	21
Himachal Pradesh	25	1,526,646	37	21	5	3	0	2	0	0	0	8	3	0	0	0	7
Jammu and Kashmir	156	7,997,349	216	11	2	6	4	0	0	0	0	1	0	2	4	0	24
Jharkhand	31	6,500,677	74	11	6	2	0	0	0	0	0	0	0	0	2	0	9
Karnataka	182	27,910,627	387	17	3	28	7	0	4	0	0	1	7	1	11	0	46
Kerala	100	12,908,553	270	0	0	20	0	0	0	0	0	1	12	2	58	0	36
Lakshadweep	4	40,405	10	0	0	0	0	0	0	0	0	0	0	0	0	0	1
Madhya Pradesh	292	35,688,704	415	12	15	9	3	5	- 1	0	0	0	0	0	20	0	48

Figure 3-14: First page of National Gap Report showing report export option

3.1.5 Help tool

The help tool in the toolbar panel generates User Manual of FDSS. The user manual contains the detailed information of FDSS to provide technical support to beginner as well as advanced level FDSS user.

3.1.6 User Management tool

The User Management tool in the Toolbar panel is only enabled when a user from Administrator group logs into the FDSS application. This tool provides an interface to the administrator to view, add, and remove users from the application. This tool also allows the administrator to change the password and the group of users from administrator to normal user or vice versa. Please refer to Section 6 where these steps are explained in detail.

3.1.7 Floating Navigation tool pane

The Floating Navigation tool pane as shown in Figure 3-15 helps the user to navigate through the map currently displayed in the Map window. Table 3-2 illustrates various buttons in this tool pane and their functions.



(Q	•

Figure 3-15: Floating Navigation toolbar

Tool icon	Tool name	Description
	Pan North	This tool pans the map in north direction. The map is navigated up when the user clicks on the tool.
	Pan West	This tool pans the map in west direction. The map is navigated to left when the user clicks on the tool.
	Pan South	This tool pans the map in south direction. The map navigates down when user clicks on the tool.
	Pan East	This tool pans the map in east direction. The map navigates to right when user clicks on the tool.
	Zoom to India	This tool zooms in the Map window to extent of India.
	Zoom In	Zooms in to the map. The map zooms in when the user clicks on the tool

FDSS

Fire Decision Support System



Tool icon	Tool name	Description
. P	Zoom Out	Zooms out of the map. The map zooms out when the user clicks on the tool.

3.2 Map window

The Map window (Figure 3-1) displays all the layers which are turned on by the user in the Map Tools window (Section 3.3). User has to check the appropriate layer in Map Tools to turn on a layer. By default, the Map window displays a Google Map layer when the user starts a session. Please refer to section 3.3.2.1 for more information on base layers which are displayed by default.

The user can navigate the Map window using simple functions mentioned in Section 3.1 (Toolbar panel) and Section 3.1.7 (Floating Navigation tool pane).

3.3 Map Tools window

The Map Tools window is displayed on the left side of the FDSS home page and it is divided into 2 parts - Layers and Legends. See Figure 3-16(Map Tools window highlighted by red rectangle).

Layers: The Layers section displays the various layers (displayed or capable of being displayed in the Map window) in a tree structure. Details of various layers in Map Tools window are given in Section 3.3.2.

Legends: Legend is a key to all the colors and symbols used in a map. The legends section describes how the user can understand the meaning of what the map colors and symbols represent. Details of the Legends are shown in Section 3.3.3.

The Map tools window is arranged in to tabs which are further divided in to directories. This type of arrangement is known as tree structure and the view is referred to as tree view. These terms are used throughout the manual. The following section helps the user in understanding how the layers and sub layers are arranged and how the user may navigate through them.



🍇 🔈 📈 🛛 🕹 ap Tools hh 🔝 🔾 0 E Layers Search Adjacent Fire Station Search Legends 200 km 200 m Base Layers \bigcirc Fire Station Sear 🏉 🗹 Google Ma 🔳 🗌 India Map • 🕑 • lazard Layers \odot Anu District Nicobar n and Nicobar Jelande Wind Zones Seismic Zones fghanistan R Fire Station Campbell Bay Fire Stat untry States and Union Territorie Gol Administrative Boundaries Phase-Wise States Pakistan earch results State Capital Districts 🖷 Mapit 📲 Clear [👳 Report Fire Station Na Municipality Town or City Phone1 Phone2 Fax India Emergency Numbe Officer In Charge Mobile Jurisdiction Mizoram Analysis dhra District Infrastructure Gap A Google 82.17773, 27.25463

Figure 3-16: Map Tools window with layers and legends

3.3.1 How to navigate the tree view

- Click the sign (=) to the left of the directory symbol of a layer to collapse the tree view or click the + sign (=) to the left of the directory symbol of a layer to expand the tree view, as required.
- In the expanded tree view, each layer has a number of checkboxes and or more + signs (□ □) or signs (□ □). An example of this is shown in the Figure 3-17 below. Click these signs to expand or collapse the tree view as relevant.



Figure 3-17: Navigating the tree view

3. To display any particular layer from the tree view, select the checkbox by clicking on it. The selected layer will be displayed in the Map window.



3.3.2 Layers

There are five group of Layers in FDSS application, namely the Base Layers, the Hazard Layers, Country layers, State and Union Territories layers, and the Administrative Boundaries layers. Since the FDSS user is likely to use all five layers simultaneously, it is important to understand what these layers constitute of and how they are related to each other.

By default (when the application first opens), the Map Tools window displays a part of the tree structure as shown in Figure 3-18. Various layers in Map Tools window are described summarily in the following sub sections.



Figure 3-18: Default tree view of layers in Map Tools window



3.3.2.1 Base Layers

The Base Layers section has two options, namely, Google Map and India Map with their respective checkboxes. The Base Layers behave differently, depending on whether the user is connected to the internet or not (that is when working on an intranet connection).

3.3.2.1.1 Base Layers When Connected to Internet

When the user is connected to the internet, both the checkboxes are displayed. The user can use one of two options given below:

1. Select the Google Map checkbox to display the google map in the Map window as shown in the Figure 3-19.



Figure 3-19: Base Layers when connected to internet

2. The display in the Map window corresponding to selecting the Google Map checkbox is shown in Figure 3-20.



Figure 3-20: FDSS - Google (Satellite) Map



3. Alternatively, user can select the India Map checkbox to display the India map in the Map window. This is same as the default view of Map window when the user is not connected to internet as described in the next sub section.

3.3.2.1.2 Base Layers when not connected to internet

When the user is not connected to the internet (is working across a local intranet) only the India Map checkbox is displayed as shown in the illustration below. By default, the India Map checkbox is selected. The user can only use the following option given below:

1. Deselect/select the India Map checkbox to hide/display the India map in the Map window as shown in Figure 3-21.



Figure 3-21: Base Layers when not connected to internet

2. The display in the Map window corresponding to selecting the India Map checkbox is shown in Figure 3-22.



Figure 3-22: FDSS- India Map



3.3.2.2 Hazard Layers

The Hazard Layers include Wind Zones, Seismic Zones and Climatic Zones. The layers show hazards in terms of their geographical spread, their intensities, and other hazard-related parameters involved. The expanded tree view of the Hazard Layers illustrates the complete menu and is shown in Figure 3-23.



Figure 3-23: Fully expanded tree view of Hazard layers

The Hazard Layers, when viewed independently, present the hazards at a country-level (India-level). While these may prove invaluable in getting an overview of the hazards, the Fire DSS user is often more concerned about location-specific hazards. This layer allows the user to drill down to specific locations and view the hazards specific to the selected location.

Details of each data layer under the hazard layers are elaborated in Section 5.1.

3.3.2.3 Country layer

There are five layers grouped in the Country layer, namely Fire Stations in India, Operational Fire Stations, Proposed Fire Stations, Other agency fire stations Area and District Fire Risk Category. The expanded tree view of the Country layer section illustrates the complete menu and is shown in Figure 3-24. For More information on Country layers, see Section 5.2.



Figure 3-24: Fully expanded tree view of Country layer

3.3.2.4 States and Union Territories layer

The States and Union Territories layer includes thirty five layers, each devoted to a single state or union territory. Under each State or Union Territory layer, there are several sub layers, namely, Major Roads, Minor Roads, Rail Network, Land Use Land Cover, Built up Area and Ideal urban fire station jurisdiction.



The expanded tree view of the States and Union Territories layer illustrates the complete menu and is shown in Figure 3-25.

🔄 😋 States and Union Territories
🗄 🦳 Andaman and Nicobar Islands
🕀 🦳 Andhra Pradesh
🕀 🦳 Arunachal Pradesh
🕀 🦳 Assam
🕀 🦳 Bihar
🕀 🦳 Chandigarh
🕀 🦳 Chhattisgarh
🕀 🧰 Dadra and Nagar Haveli
🕀 🧰 Daman and Diu
👜 🧰 Delhi
🕀 🧰 Goa
🕀 🧰 Gujarat
🕀 🧰 Haryana
🕀 🧰 Himachal Pradesh
🔄 🧰 Jammu and Kashmir
🕀 🧰 Jharkhand
🕀 🧰 Karnataka
🕀 🧰 Kerala
🕀 🧰 Lakshadweep
🕀 🧰 Madhya Pradesh
🕀 🧰 Maharashtra
🕀 🧰 Manipur
🕀 🧰 Meghalaya
🕀 🧰 Mizoram
🕀 🧰 Nagaland
🕀 🧰 Orissa
Puducherry
🕀 🧰 Punjab
🕀 🧰 Rajasthan
🕀 🧰 Sikkim
🕀 🧰 Tamil Nadu
🕀 🧰 Tripura
🕀 🧰 Uttar Pradesh
Uttarakhand
🕁 🧰 West Bengal

Figure 3-25: Expanded tree view of States and Union Territories

For more information on the States and Union Territories layer see Section 5.3.


3.3.2.5 Administrative Boundaries layer

The Administrative Boundaries layer includes, Phase-Wise States, State Capital and Districts layer. The expanded tree view of the Administrative Boundaries layer illustrates the complete menu and is shown in Figure 3-26.

Administrative Boundaries	
E State Capital	
Districts	

Figure 3-26: Expanded tree view of the Administrative Boundaries layer

For more information on the Administrative Boundaries layer see Section 5.4.

3.3.3 Legends overview

The map displayed in Map window displays information using symbols. These symbols can be figures, shapes, lines and colors that represent quantity / type/ classification present in the attribute database of a feature (GIS data) that is being presented in the window. Therefore, the map's legend tells the user what these symbols mean. This enables the user to interpret the map displayed and draw inferences.

Each layer in FDSS application is essentially a map. Therefore, each layer has its own unique legend. In the FDSS User Interface, user has the provision to view legend of the same layer in three ways.

- a. Legend of a specific layer is displayed when the user points⁴ the hand pointer \sum over that layer.
- b. Legend of the same layer is also displayed in Search window (Section 3.4) under the Legends tab. This display is dynamic in nature and is triggered when a user checks a layer. This functionality is described in detail in Section 3.4.3.
- c. Legend of the same layer can also be viewed in Map Tools window as described under.

The Legends tab is hidden by default when a user logs into an FDSS session (See Figure 3-18). User has to expand the tab by clicking on expand icon in (highlighted by blue box in

⁴ Pointing the mouse is also called Hover or Rollover. User has to only "point" the mouse over the layer. Clicking on the layer does not display any legend.



Figure 3-18). When the tab is expanded, user can see a master list ⁵of legends arranged in the order of their occurrence in Layers tab (described in Section 3.3.2).

Figure 3-27 illustrates a sample legend view in the Map tools window when the user expands the legend tab and scrolls the seeker bar (highlighted in red box) down to view the legends.



Figure 3-27: Expanded Legends tab in Map Tools window showing Seeker bar

3.4 Search window

The Search window appears on the right side of the FDSS homepage. There are five search options arranged in tabs in this window. They are listed as under:

- 1. Search
- 2. Adjacent Fire Station Search
- 3. Legends
- 4. Select by Attributes
- 5. Select by Location

⁵ The legend of all the layers are listed in the Legends tab. There is no group wise arrangement. User has to scroll down to the legend of the appropriate layer in the list to view the legend.



These search options are illustrated in Figure 3-28. User can move the seeker bar provided at the bottom of the window in order to view relevant tab.

1	2		3	4	5 (
earch	Adjacent Fire St	ation Search	Legends	Select by Attributes	Select by Locat
ire Sta	tion Search				
ne stu	and a contain				
State		-			
	man and Nicobar Is	ilands 💌			
District			à		
Fire Sta			•		
	obell Bay Fire Statio	n 💌			
Go!					
Go!					
	results				
📕 Map	it 🗏 Clear [R	eport			
Fire St	ation Name				
Municip					
Town					
Phone					
Phone	2				
Fax					
Emerge	ency Number				
Officer	In Charge				
Mobile					
Jurisdi	ction				
Analysis	5				
State					
	man and Nicobar Is	lands 💌			
District	2		a		
-	1 1	- 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 110 - 1	•		
Infras	structure	ap Analysis			
Mov	e the seel	er bar	to view	more search	options
			V		aptions
-					1
			111		

Figure 3-28: Search window displaying search options

3.4.1 Search

This is the first tab in the Search window (Figure 3-28). The main functionality of the Search option is to search for a particular fire station in India by district as specified by the user. Besides that, the Fire Station Search as shown in Figure 3-29, allows users to carry out the following functions:

- Select a State or Union Territory
- Select a district from the State or UT selected above
- Zoom to the selected district in Map window
- Select a particular fire station from the district selected above
- Map the selected fire station in the Map window
- Generate a report for the selected fire station



ire Station Search	
State	
Andaman and Nicoba	ar Islands 💌
District	~
Nicobar	
Fire Station	
Campbell Bay Fire St	ation
earch results	
🖷 Mapit 📲 Clear [Report
🖶 Mapit 📲 Clear [Fire Station Name Municipality	Report
Imapit 📕 Clear [Fire Station Name Municipality Town or City	Report
Mapit R Clear	Report
Imapit 📕 Clear [Fire Station Name Municipality Town or City	Report
Mapit R Clear	Report
Fire Station Name Municipality Town or City Phone 1 Phone 2	Report
Fire Station Name Municipality Town or City Phone1 Phone2 Fax	Report

Figure 3-29: Fire Station Search functionality

3.4.1.1 Zoom into a selected district within a state

This functionality allows the user to zoom into any district within the state by following the steps mentioned below:

- 1. Select the state name from the State dropdown box. This will display the names of all the districts within that state in the District dropdown box.
- 2. Select the district of choice within the state from the District dropdown box.
- 3. Click the button highlighted by a red circle in Figure 3-29. This tool zooms to the selected district and displays it in the Map window.

Figure 3-30 shows the result of a search made for the Central Delhi district within the Delhi state and the resultant view in the Map window.





Figure 3-30: Zoom into a selected district

Notes:

- The zoom function in the fire station Search window only changes the display in the Map window.
- User cannot use the Go button, the controls in the Search results frame (namely, Map It, Clear and Report buttons) with the zoom into a district search function. Also the table in the Search results frame is not populated when the zoom into a district search function is used.

3.4.1.2 Select and Map a Fire Station

The Fire Station Search allows a user to select a particular fire station from a particular district within a state by following the steps mentioned below:

- Select the state, district and fire station from the State, District, and Fire Station dropdowns respectively as shown in Figure 3-30. As an example, the Connaught Circus fire station of Central Delhi district in Delhi state has been shown in Figure 3-31.
- 2. Click the Go button (as shown by the blue highlighted box in Figure 3-31).
- 3. The Search results table (as shown by the green highlighted box in Figure 3-31) displays all the information for the selected fire station.



re Station Search	
State	
Delhi	
District	
Central	
Fire Station	
Cannaught Circus Fire	e Station 💌
arch results Maph 🖫 Clear 🧊	Report
Fire Station Name	Cannaught Circus Fire Station
Fire Station Name Municipality	Cannaught Circus Fire Station
Fire Station Name Municipality Town or City	Cannaught Circus Fire Station Ndmc Delhi
Fire Station Name Municipality Town or City Phone 1	Cannaught Circus Fire Station Ndmc Delhi 01123414444
Fire Station Name Municipality Town or City Phone 1 Phone 2	Cannaught Circus Fire Station Ndmc Delhi 01123414444 Not Available
Fire Station Name Municipality Town or City Phone 1 Phone 2 Fax	Cannaught Circus Fire Station Ndmc Delhi 01123414444 Not Available Not Available
Fire Station Name Municipality Town or City Phone1 Phone2 Fax Emergency Number	Cannaught Circus Fire Station Ndmc Delhi 01123414444 Not Available Not Available 101
Fire Station Name Municipality Town or City Phone 1 Phone 2 Fax Emergency Number Officer In Charge Mobile	Cannaught Circus Fire Station Ndmc Delhi 01123414444 Not Available Not Available

Figure 3-31: Search results table showing Map It, Clear and Report options

To map the fire station, click the Map It button (as shown by the red highlighted box in Figure 3-31). The Map window is zoomed to the Connaught Circus Fire Station with a place marker indicating its position as shown in Figure 3-32.



Figure 3-32: Fire Station search showing populated table and corresponding result mapped in the Map window after using the Map It button

- 5. To remove the place marker if from the Map window, click Clear button (As shown by the orange highlighted box in Figure 3-31) in the Search results frame inside the Search window.
- 6. Click the Report button (As shown by the tan highlighted box in Figure 3-31) in the Search results section in the Search window to generate a fire station report for the searched fire station. The fire station report is displayed in a new window and a sample report is shown in Figure 3-33.



FDSS



Fire Decision Support System

14 4 1 of 11 ▶ ▶1	Select a format Export	۵
		Directorate General NDRF & CD (Fire)
		Ministry of Home Affairs, Govt. of India
5DSS		
Fire Station Report:	Cannaught Circus Fir	e Station
Fire Station General Inform	nation	
State District	: Delhi : Central	Participant Sa
Fire Station Name	: Cannaught Circus Fire Station	and the second s
Latitude/Longitude	: 28°37.92' N, 77°13.4' E	MRy HBlock Commy New Delhi
Address	: Cannaught Circus Fire Station New Delhi Pin Code-110001	Shivaji Chuk
Under Administration of (Divisional Office)	: Delhi Fire Service Head Quarters	Rajiv Chowk Rajiv Chowk
Divisional Officer Name and Designation	: G.C. Mishra, C.F.O	Metro Station M Metro Station
Total Fire Stations Fall under Administration	: 52	Subar Connaugh Cir Road Metro
Officer-in-Charge (On Duty)	: Mukesh Verma	
Designation	: Station Officer	
Mobile No. of Officer-in-Charge	: 9810445332	Mantar 2 Control Past Ch ob Manba Ray
Fire Station Phone Numbers	: 01123414444	Concest as a second sec
Emergency Fire Phone Numbers	: 101	
Fire Station is under the Administration of	: State Goverment	
Fire Station Surveyed By	: RMSI Personnel	
Date of Survey	: August, 2011	

Figure 3-33: Sample of a searched fire station report

3.4.1.2.1 Exporting the fire station search report:

Please follow the instructions in Section 3.1.1.1 (Exporting and Printing a Generated Map or Report) to Export and Print the Fire Station Search Report.

3.4.2 Adjacent Fire Station Search

This is the second tab in the Search window (Figure 3-28). The Adjacent Fire Station Search functionality (shown by the blue highlighted box in Figure 3-34) helps the user in searching for all fire stations within a user specified distance from a particular fire station. This is particularly helpful for decision making in the event of a fire incident. A fire station can request back up (depending upon extent of fire) from nearby fire stations based on their proximity from the fire incident.

To operate this tool, the user has to follow the steps mentioned below:

- 1. Select the state from State dropdown box.
- 2. Districts from the selected state are populated in the District dropdown box. User has to select a district from the District dropdown.





- 3. Fire Stations under the selected district are populated in the Fire Station dropdown box. User has to select a fire station from this dropdown. As an example, Connaught Circus Fire Station in Central district of Delhi state is shown in Figure 3-34.
- 4. Next, the user types in a distance⁶ in the Radius textbox (Shown by the red highlighted box in Figure 3-34) to identify fire stations in the proximity.
- 5. Click on Gol button. This will perform two tasks. Firstly, it will calculate the aerial distance and displays a circular boundary of specified radius with the selected fire station at the center of the circle. Secondly, it will generate an Adjacent Fire Station search report containing information of all the fire stations falling within the user specified radius. Figure 3-34 and Figure 3-35 shows all the fire stations falling within the specified radius (10 km in this example) from Connaught Circus fire station and a sample report of it respectively.
- 6. The report generated can be exported using instructions mentioned in Section 3.1.1.1 (Exporting and Printing a Generated Map or Report).
- 7. Click on Clear button to clear the fire station results and circular boundary from the Map window display.



Figure 3-34: Adjacent Fire Station Search window

⁶ Unit of radius is Kilometers (Km)



FD	ss						istry of Home A	
djacent Fir	e Stations to 'Canna	ught Circus Fire Statio	n' within 10	km radius				
Aerial Distance(km)	Fire Station	Street Address	Municipality	Officer in Charge Name	Officer in Charge Mobile No	Phone	District	State
0.10	DFS Headquarter Fire Station	Cannaught Place New Delhi Pin Code-110001	NDMC	Mr. Mahmood	9968306650	01123412222	Central	Delhi
1.94	Rakabganj Fire Station	Rakabganj Fire Station New Delhi Pin Code-110001	NDMC	Hari Chand	9810496802	01123719497	Central	Delhi
2.43	Paiwalan Fire Station	Paiwalan Fire Station Near Jama Masjid Delhi,Pin Code-110006	MCD	Tarkeswar Rai	9868756761	01123244101	New Delhi	Delhi
2.53	Darya Ganj Fire Station	Darya Ganj Fire Station New Delhi Pin Code-110002	MCD	Bacchanlal	9868124930	01123268219	New Delhi	Delhi
2.81	Delhi Secretariat Fire Station	Delhi Secretariat,New Delhi Pin Code-110002	MCD	Rajpal Singh	9953402991	01123392677	New Delhi	Delhi
2.93	Rashtrapati Bhawan Fire Station	Rashtrapati Bhawan New Delhi Pin Code-110001	NDMC	Ved Prakash	9818767080	011-23015321	Central	Delhi
3.13	Teliwara Fire Staion	Qutub Road,Sadar Bazar Delhi, Pin Code-110006	MCD	Manish Kumar	9891987763	01123612755	North	Delhi
3.29	Rani Jhansi Road Fire Station	Filmistan Cinema, Rani Jhansi Road, Delhi - 110005	MCD	Avtar Singh	9811881761	01123672229	New Delhi	Delhi
3.30	S P Mukherjee Marg Fire Station	S P Mukherjee Marg,Delhi Pin Code-110006	MCD	Sanjay Tomar		01123975553	North	Delhi
4.40	Prasad Nagar Fire Station	Prasad Nagar,Delhi Pin Code-110008	MCD	N.P. Sharma	9818031656	01125751999	New Delhi	Delhi
4.56	Chanakyapuri Fire Station	Chanakyapuri Fire Station Kautilya Marg,New Delhi Pin Code-110021	NDMC	Rajbir Kharb	9968272496	01126112226	Central	Delhi
4.75	Shanker Road Fire Station	Shankar Road New Delhi Pin Code-110008	MCD	Mr. Roop Chandra	9718093701	011287422	New Delhi	Delhi
5.12	Safdarjung Fire Station	Safdarjung Fire Station,Jor Bagh, New Delhi Pin Code-110003	MCD	S S Yadav	101	01124611111	South	Delhi
5.16	Geeta Colony Fire Station	Near St. Lawrence School Geeta Colony,Delhi Pin Code- 110031	MCD	Mr. Ved Pal Chikara	9868202979	01122513330	East	Delhi
6.16	Laxmi Nagar Fire Station	Near Scope Minar,Laxmi Nagar Delhi,Pin Code-110092	MCD	Nagendra Singh	9868882111	01122417747	East	Delhi
6.19	Shashtri Park Fire Station	Near Jag Parvez Chand hospital Shashtri Park Pin Code-110053	MCD	Udai Vir Singh Tomar	9868154496	01122853933	North East	Delhi
6.23	Roop Nagar Fire Station	Roop Nagar,Delhi Pin Code-110007	MCD	Mr. K.K. Saxena	9810228404	01123842505	North	Delhi
7.40	Mathura Road Fire Station	Ashram Chowk,Delhi Pin	MCD	Bharat Singh		01126341301	South	Delhi

Figure 3-35: Sample report of Adjacent Fire Station Search tool

3.4.3 Legends

As mentioned in Section 3.3.3 each layer in the FDSS application is essentially a map. Therefore, each layer has its own unique legend. In the FDSS User Interface, Legends can also be viewed using the tab provided in the Search window⁷.

As described in Section 3.2 (Map window) and Section 3.3.2 (Layers), a layer is turned on when the user checks the appropriate checkbox for that layer. The layer is also displayed in the Map window. In addition to that, the legend corresponding to the checked layer is displayed in the Legends tab in the Search window.

To view the legend of a specific layer in the Legends tab in Search window, the user has to follow the steps described below:

⁷ Legends tab is numbered 3 in Figure 3-28



- 1. Check the layer in the Map Tools window (For example, Wind Zones layer as shown by red highlighted box in Figure 3-36 is selected).
- 2. Click on the Legends tab in the Search window (Shown by the orange highlighted box in Figure 3-36). The Legends tab will display the legend of the selected layer in the Layers tab.

As an example, the legend of the Wind-Zones layer under Hazard Layers is shown in Figure 3-36.



Figure 3-36: Legends tab in Search window illustrating legend of wind zones

3.4.4 Select by Attributes

The Select by Attributes tab provides users with the facility to map or highlight an active layer based on the attributes of the layer. To operate this tool, the user has to follow the steps mentioned below:

1. Click on the 'Select by Attributes' tab provided in the Search window to open the Select by Attributes table as shown in Figure 3-37.



Search A	djacent Fire Station Search	Legends	Select by Attributes
Layer Info	rmation		
Map:	Select layer		~
Operation	i.		
Method:	Select method		~
Query Con	dtion		
Columns			
No Columns	available		
Condition:	~		
Value:		*	AND OR
			~
			~
			Ok

Figure 3-37: User Interface of Select by Attribute tool

2. Activate the layer or layers by clicking the checkbox or checkboxes in the Map Tools window for which information is required using the Select by Attributes tool. Only the layers that are activated in the Map Tools window will appear in the Map dropdown box of the Select by Attributes tab. As an example Wind Zones and Seismic Zones layers selected in Map Tools window (highlighted by the red rectangle in Figure 3-38) get populated in the Map dropdown (highlighted by the green rectangle in Figure 3-38).



Figure 3-38: Layer information generated in Map dropdown

- 3. Fill in the required information from the relevant dropdown boxes as follows:
 - A. Map (in Layer Information) Select one of the currently activated layers to be mapped. As an example, Seismic Zone layer is selected.
 - B. Method (in Operation) The application allows the user to opt for any of the three methods viz.
 - Create a new selection: This method is opted when a user wishes to create a new query. In the example, 'Create a new selection' is chosen
 - Add to a current selection: 'Add to current selection' is used to map the already mapped layer with some other set of attributes This has to be opted in combination with the previously executed query (created using 'Create a new selection' method). When a query is executed using this method, it displays both the outputs generated using 'Create a new selection' method and 'Add to a current selection' method.
 - Remove from current selection: This method is opted when a user wish to clear the output (highlighted area) generated on the Map window.
 - C. Columns (in Query Condition) Columns in the Query Condition frame contains the list of attributes of the selected layer. In the example, the column for which query needs to be executed (Here, seismic zone column/attribute is selected, highlighted by inner green rectangle in Figure 3-39).





- D. Condition (in Query Condition) The Condition dropdown contains various relational information (Here = has been selected). Details of the conditions are given in Table 3-3.
- E. Value (in Query Condition) This displays the information available in the database for the selected column. Select the value from the Value dropdown (Here Zone II has been selected).
- 4. Click the AND button (as highlighted by the red rectangle in the figure below) to write the information provided in the Query Condition in a query format in the textbox provided below (highlighted by blue rectangle).



Figure 3-39: Example showing use of Select by Attribute tool

5. Click the Ok button (as highlighted by the pink rectangle in the figure above). This highlights the area which lies within Zone II in the Map window as shown in Figure 3-40.



Figure 3-40: Map showing area lying under Zone II after unchecking the checkbox corresponding to Seismic Zones layer

6. In order to remove the selection, select the method 'Remove from current selection' and click on the Ok button. This deselects the highlighted area.

Details of each condition that can be selected in the Select by Attributes tab are given in Table 3-3:

Condition	Detail
=	Equal To
>	Greater Than

Table 3-3: Details of conditions in the Select by Attributes tool





Condition	Detail
>=	Greater Than Equal To
<	Less Than
<=	Less Than Equal To
<>	Not Equal To
Like	Search for specified characters in column

3.4.5 Select by Location

The Select by Location functionality provides users the facility to map or highlight an active layer based on its location attributes with respect to another active layer. To operate this tool, the user has to follow the steps mentioned below:

1. Click on the Select by Location tab provided in the Search window (numbered 5 in Figure 3-28). The Select by Location tab is displayed as shown in Figure 3-41.



Search Adjacent Fire	Station Search	Legends Select by	/ Attributes	Select by Locatio
Layer Information				
Select features from:	Select layer		~	
that:	Select operation	on	~	
features in the layer:	Select layer		~	
Apply a buffer of:	0.000000	Select Degree	~	
		ОК Н	ide	

Figure 3-41 : User Interface for Select by Location tab

2. Activate the layers on which the Select by Location function is to be run by clicking the checkbox besides it in the Map Tools window. Only the layers that are activated in the Map Tools window will appear in the Select by Attributes tab. As an example Operational Fire Stations and Proposed Fire Stations layers selected (as highlighted by red rectangle in Figure 3-42) get populated in the 'Select features from' dropdown. (Highlighted by green rectangle in the Figure 3-42).





Figure 3-42: Activating the layer

- 3. Fill in the required information from the relevant dropdown boxes of the 'Select by Location' tab as follows: (See Figure 3-43).
 - A. Select features from The activated layer from which features will be selected (Operational Fire Stations layer is selected in the example).
 - B. That the spatial function to be assessed ('are within a distance of' is selected in the example).
 - C. Features in the layer The activated layer with respect to which the spatial relationship of the previous layer will be assessed (Proposed Fire Stations is selected in the example).
 - D. Apply a buffer of The buffer distance (10 is selected in the example and the unit is taken as Kilometers).



Figure 3-43: Filling layer information

4. Now Click the OK button (highlighted by the blue rectangle in Figure 3-43) to highlight all the operational fire stations in India which are at a distance of 10 Kilometers from the proposed fire stations in India on the map in the Map window. The resultant map is shown in Figure 3-44.







Figure 3-44 : Map showing operational fire stations that are within a distance of 10 km from proposed fire stations

5. FDSS provides options of spatial functions that can be analyzed from the Select by Location tab. Details of each spatial function that can be selected in the Select by Location tab are given in Table 3-4 below:

Spatial Function	Detail			
Intersect	Returns true if geometry a shares any portion in space wit geometry B			
Are within a distance of	Returns true if geometry A is within a specified distance of geometry B			

Table 3-4: Detai	s of spatia	I functions in	Select by	I ocation tool
10010 0 11 00101	o or opada		00.000.0	



Spatial Function	Detail
Completely contain	Returns true if geometry B intersects the interior of A but not the boundary
Are completely within	Returns true if geometry A is completely inside geometry B
Have their centroid in	Returns true if the centroid of geometry A lies inside the geometry B
Touch the boundary of	Returns true if the geometries have at least one point in common but their interiors do not intersect
Are identical to	Returns true if A and B have the same geometry
Are crossed by outline of	Returns true if the geometries have some but not all interior points in common
Contain	Returns true if and only if no points of geometry B lie in the exterior of A and at least one point of interior B lies inside A
Are contained by	Returns true if no point in geometry A is outside geometry B

Notes:

- Geometry A here refers to the layer selected in the Select features from dropdown box.
- Geometry B here refers to the layer selected in 'features in the layer' dropdown box



3.5 Analysis window

The Analysis window is located at the bottom right corner of the FDSS User Interface. There are two types of analysis tools available in this window namely, Infrastructure and Gap Analysis as shown in Figure 3-45.



Figure 3-45: Analysis window showing analysis tools

The analysis tools allow users to generate Infrastructure and Gap Analysis reports by using the corresponding buttons located at the bottom of it. These reports can be generated at both the district and state levels.

3.5.1 Infrastructure

The Infrastructure tool is shown by green highlighted box in Figure 3-45. The Infrastructure tool helps in identifying the available trained manpower, specialized equipment, and fire fighting vehicles for all the fire stations in a particular state or in a particular district of a state. As the Infrastructure report is a summation of details for all the fire stations, user can view infrastructure details of individual fire stations as well.

3.5.1.1 Generating an Infrastructure report

- 1. Select a state from the State dropdown list. As an example Delhi state is selected in Figure 3-46.
- 2. Select a district from the District dropdown list. In this dropdown list, user has two options:



- a. **Select All option from District dropdown**: This option generates State Infrastructure Report for the state selected in Step 1 as all the districts are selected. This option is highlighted with a green box in Figure 3-46.
- b. **Select a particular district from District dropdown**: This option generates District Infrastructure report for the selected district. This option is highlighted with a blue box in Figure 3-46.
- 3. The Zoom icon (highlighted by purple circle in Figure 3-45 and Figure 3-46) displays the selected district or all districts in the Map window.
- 4. Click the Infrastructure button (highlighted in green box in Figure 3-45). This opens a new browser page showing the Infrastructure report as selected by the user.
- 5. To export or print the report, follow instructions provided in Section- 3.1.1.1.

Analysis	
State	
Delhi	
District	0
Central	
All	
Central	
East	
New Delhi	
North	
North East	
North West	
South	
South West	
West	

Figure 3-46: Analysis window illustrating State and District selections

A sample page of State Infrastructure Report for all districts in Delhi State is shown in Figure 3-47.

FDSS



Fire Decision Support System

i4 4 1 of 13 ▶ ▶i	Select a format 👻 Export	
2		Directorate General NDRF & CD (Fire)
		Ministry of Home Affairs, Govt. of India
FDSS		١
State Infrastructure Re	eport: Delhi	
State	: Delhi	A Mal I
State Headquarter Location	: Delhi Fire Service Head Quarters	Detrike Man
Name of State Head Quater Officer	: G.C. Mishra (C.F.O) e Country" with an Objective to Prepare Capital Investment a	Image: State of the state
Fire Services in the Country	4 -540	
	1 of 13	

Figure 3-47: Sample page from Infrastructure report for Delhi State (All Districts selected)

A sample page of District Infrastructure Report for Central district in Delhi State is shown in Figure 3-48.

FDSS



Fire Decision Support System



Figure 3-48: Sample page from Infrastructure report of Central district, Delhi state (Only Central district selected)

3.5.2 Gap Analysis

Gap Analysis tool identifies additional requirement to be added upon the existing resources mentioned in Infrastructure reports (Section 3.5.1). Gap Analysis tool helps the user to generate a report showing the gaps between available and required specialized equipment, trained manpower, and fire fighting vehicles for the fire stations in a selected district or all districts in a particular state. As the Gap Analysis report is a summation of gaps for all the fire stations, user can view gap details ire stations at district level as well.



3.5.2.1 Generating a Gap Analysis Report

The interface for both Gap Analysis tool and Infrastructure tool described earlier in Section 3.5.1 is same. As a result, the instructions for generating Gap analysis report are similar. These steps are listed as under:

- 1. Select a state from the State dropdown list. As an example Delhi state is selected in Figure 3-46.
- 2. Select a district from the District dropdown list. In this dropdown list, user has two options:
 - a. **Select All option from a District dropdown**: This option generates state gap analysis report for the state selected in Step 1 as all the districts are selected. This option is highlighted with a green box in Figure 3-46.
 - b. Select a particular district from District dropdown: This option generates district gap analysis report for the selected district. This option is highlighted with a blue box in Figure 3-46.
- 3. The Zoom icon (highlighted by purple circle in Figure 3-45 and Figure 3-46) displays the selected district or all districts in the Map window.
- 4. Click the Gap Analysis button (highlighted in blue box in Figure 3-45). This opens a new browser page showing the Gap Analysis report as selected by the user.
- 5. To export or print the report, follow instructions provided in Section- 3.1.1.1.

A sample page of state gap analysis report for all districts in Delhi state is shown in Figure 3-49.



re Dec	cisior	n Suppo	ort Sy	/ster	n										Deliveri	ng a worl	d of solution
14 4 1	of 32	Se Se	lect a form	at 👻 E	xport	٢								D	irectorate G	eneral N	IDRF & CD (
														784	Ministr	y of Home	Affairs, Govt. o
💹 FD	SS																0
					9	Gap Ana	alysis	Repo	ort for	State	e Dell	ni					
/ehicle Deta	ails																
Current Veh	icle Invent	tory for Operati	onal Fire	Stations													
												0.000		-			*****
District	Fire Stations	Ideally Served Population Estimates	Water Tenders	Water Bowsers	Foam Tenders	Advanced Rescue Responders	Sky Lifts / TTL	DCP Tenders	Hose Tenders	BA Vans	Hazmat Vans	QRT	Motor Cycle Mists	Fire Boats	Ambu-lances	Education Varis	Total Vehicle
Central	5	196,309	6	5	1	0	2	0	0	1	0	1	1	0	0	0	
East	6	1,747,840	11	6	1	2	0	1	0	1	1	1	2	0	0	0	
New Delhi	6	658,649	9	4	0	1	2	0	0	0	0	0	2	0	0	0	
North	3	458,152	5	2	1	0	0	0	1	0	0	1	2	0	0	0	
North East	3	580,798	4	3	0	0	0	0	1	0	0	0	0	0	0	0	
North West	9	1,042,597	13	7	1	1	1	0	0	0	1	1	1	0	0	0	
South	6	778,384	10	6	2	2	1	1	া	1	1	2	1	0	0	0	
South West	5	1,009,772	6	9	0	0	0	0	0	1	0	0	1	0	0	0	
West	10	1,654,111	15	6	0	1	0	2	2	0	0	1	0	0	0	0	
Total	53	8,126,612	79	48	6	7	6	4	5	4	3	7	10	0	0	0	1
Vehicle Gap	in Operat	ional Fire Stati	ons														
District	Fire Stations	Ideally Served Population Estimates	Water Tenders	Water Bowsers	Foam Tenders	Advanced Rescue Responders	Sky Lifts /TTL	DCP Tenders	Hose Tenders	BA Vans	Hazmat Vans	QRT	Motor Cycle Mists	Fire Boats	Ambu-lances	Education Vans	Total Vehicle
Central	5	196,309	-1	-5	0	1	-1	1	1	1	0	-1	-1	0	0	1	
East	6	1,747,840	5	7	3	1	2	1	2	1	0	4	3	0	0	2	
New Delhi	6	658,649	-1	1	2	1	0	1	2	2	0	3	1	0	0	2	
North	3	458,152	0	0	1	0	0	0	1	0	0	0	-1	0	0	0	1
North East	3	580,798	1	1	2	0	0	0	-1	0	0	2	2	0	0	0	
North West	9	1,042,597	0	-4	5	0	0	1	1	1	0	1	1	0	0	0	
South	6	778,384	-3	-3	3	0	0	2	0	0	0		-1		0	0	
South West	5	1,009,772	3	-1	4	0	1	0		0	- 22		0	0.	0	1	
West	10	1,654,111	1	4	7	2	0	1	-1	1	0		0		0	1	
Total	53	8,126,612	5	0	27	5	2	7	3	6	0	6	4	0	0	7	3

Figure 3-49: Sample page from Gap Analysis report for Delhi state (All districts selected)



A sample page of district Gap Analysis Report for Central district in Delhi State is shown in Figure 3-50.

															Directorate Mini	General I stry of Home	
FD:	ss																(
					Gap	Analy	sis R	eport	for D	istric	Cent	ral(D	elhi)				
ehicle Deta	ails																
District	Fire Stations	Ideally Served Population Estimates	Water Tender	Water Bowser	Foam Tender	Advanced Rescue Responder	Sky Lift/ TTL	DCP Tender	Hose Tender	BA Van	Hazmat Van	QRT	Motor Cycle Mist	Fire Boat	Ambu-lance	Education Van	Total Vehicles
Current Vehic		-															
Central	5	196309	6	5	1	0	2	0	0	1	0	1	1	0	0	0	17
Vehicle Gap in	Operationa	I Fire Station	15														
Central	5	196309	-1	-5	0	1	-1	1	1	1	0	-1	-1	0	0	1	-4
Additional Veh	nicles requir	red for New I	Jrban Fire S	Stations													
Central	1	39774	1	0	0	0	0	0	0	0	0	0	0	0	0	0	1
Total Vehicle (21	
Central	6	236083	0	-5	0	1	-1	1	1	1	0	-1	-1	0	0	1	-3
Additional Veh	nicles requir	ed for New I	Rural Fire S	tations													
Central	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
ehicle Cos	t In Lakh	Rupee															
District	Fire Stations	Ideally Served Population	Water Tender	Water Bowser	Foam Tender	Advanced Rescue Responder	Sky Lift/ TTL	DCP Tender	Hose Tender	BA Van	Hazmat Van	QRT	Motor Cycle Mist	Fire Boat	Ambu-lance	Education Van	Total Vehicle cost
Value of Curre	ent Vehicle I	Estimates nventory for	Operationa	al Fire Statio	ons												
Central	5	196309	210.00	150.00	40.00	0.00	1,000.00	0.00	0.00	30.00	0.00	9.00	6.75	0.00	0.00	0.00	1,445.75
Cost of Vehicl	1	and the second	0.04073		100	a an		100	0.10	1000			200	0.20			
	5	196309	-35.00	-150.00	0.00	500.00	-500.00	35.00	30.00	30.00	0.00	-9.00	-6.75	0.00	0.00	20.00	-85.75

Figure 3-50: Sample page from Gap Analysis report of Central district, Delhi state (Only Central district selected)

3.6 Results window

The Results window only works in conjunction with the Identify a feature tool provided in the tool bar panel (Described in Section 3.1) and, therefore, their combined use is described here in detail.

The Results window only displays information corresponding to a specific location in the Map window (selected by using the Identify a feature tool) and the layer selected in the Map Tools



window. In other words, if the user clicks on a certain location in the Map window using the Identify a feature tool after selecting the Seismic Zones layer in the Map Tools window, the Results window will only show seismic zone information related to the location where the user clicked. This information is populated from the FDSS database.

By default, this window is hidden when the user first logs into a session. The Results window can be displayed by using the Unhide 🔊 button visible on the extreme right below the Analysis tool. Conversely, when displayed, the window can be hidden by clicking the Hide 😒 button visible on the extreme right below the Analysis tool. Figure 3-51 shows the Hide button as an example.

To use the Results window and the Identify a feature tool follow the steps described below:

- 1. Click on the Identify a feature tool.
- 2. Click on the label of the layer whose information needs to be displayed.
- 3. Click on the specific location in the Map window.
- 4. The Results window will automatically open at the bottom of the screen displaying available information for the selected location and selected layer as shown in Figure 3-51.

In the example given in Figure 3-51, the user has first selected the Identify a feature tool in the Toolbar, then clicked on the Operational Fire Stations layer in the Map Tools window and finally clicked on Bikaner District (in Rajasthan) in the Map window. The information corresponding to the layer and the geographical location are displayed in the Results window.

The user can hide the Results window by clicking the Hide 💌 button just below the Analysis tool as shown in Figure 3-51 (highlighted by green rectangle).

FDSS



Fire Decision Support System



Figure 3-51: Results window showing the populated results with the use of Identify a feature tool

3.7 Map Elements

In the FDSS User Interface, there are additional elements which are not used in any type of analysis or reporting. They are solely used for the cartographic representation of map displayed in the Map window. They are called as Map elements and are shown in Figure 3-52.

There are two map elements in the FDSS user Interface.

- a. Scale Bar
- b. Map Co-ordinates

Google



Figure 3-52: Map elements in FDSS User Interface

81.65039, 33.52308

Map Co-ordinates





4 Viewing Hazard Zones, and Fire Risk

FDSS allows users to view the distribution of Wind, Seismic, and Climatic Zones across India (country level) or to determine their status at a specific geographic location (state or district level, i.e., location-specific) within India. The sections below take the user through the steps to view these zones at the country level and Fire risk at district level.

4.1 Viewing Hazard Zones at the Country Level

At this level, the user can view the distribution of the Wind, Seismic, and Climatic Zones across India.

- 1. Click the Zoom to India icon in the Floating Navigation tool pane to zoom to the country boundary extent. This step may not be necessary when the user first logs into a session since the default Map window view is set to the country boundary extents.
- 2. Click the + sign (=) to the left of the directory symbol of the Hazard Layer (Refer to area highlighted by red rectangle in Figure 4-1). This expands the tree view to display the Wind zones, Seismic zones, and Climatic zones layers.
- 3. Select the checkbox or checkboxes ☑ next to the layer or layers to be viewed in the Map window. The illustration in Figure 4-1 shows the mapping of the Seismic zone layer. Users can select one or multiple layers at the same time.



Figure 4-1: Viewing levels of seismic hazard risk (Seismic zones) at country level



4.2 Viewing District Fire Risk at the Country-Level

FDSS categorizes each district in India as per a Fire Risk Index, which has 4 categories, namely Very High, High, Medium and Low (fire risk). The user can view the district fire risk category at the country-level by following the steps below:

- 1. Click the Zoom to India icon in the Floating Navigation tool pane to zoom to the country boundary extent. This step may not be necessary when the user first logs into a session since the default Map window view is set to the country boundary extents.
- 2. Click the + sign ([⊥]) to the left of the directory symbol of the Country Layer. This expands the tree view to display the District Fire Risk Category checkbox (Refer to area highlighted by red rectangle in Figure 4-2).
- 3. Select the District Fire Risk Category checkbox to display the District level fire risk in all the states, in the Map window. The illustration in Figure 4-2 shows the mapping of the District Fire Risk Category layer (at a country-level) and its corresponding Legend popup.



Figure 4-2: Map showing district fire risk category in India



5 Layers

While Section 4 assists the FDSS user in performing various key functions through the system and explains the procedures to be followed in a step-wise manner. This section provides a description of all the layers and sub-layers present in the Map Tools window of FDSS.

5.1 Hazard Layers

This section describes the various sub-layers under the Hazard layer. A fully expanded view of the hazard layers' tree structure is shown in Figure 5-1



Figure 5-1: Tree Structure of the Hazard Layers

5.1.1 Wind Zones

The Wind Zones layer classifies India into different wind zones based on the intensity of the wind hazard, which is directly correlated to the wind velocity. The Wind Zones layer is displayed on selecting the Wind Zones checkbox. The classification and the corresponding Legend popup are shown in Figure 5-2.



Figure 5-2: Map showing wind zones in India



5.1.2 Seismic Zones

The whole country is classified into 4 Seismic zones. To view which part of India lies under which seismic zone select the checkbox corresponding to the Seismic Zones layer. The classification and the corresponding legend popup are shown in Figure 5-3.



Figure 5-3: Map showing seismic zones in India

5.1.3 Climatic Zones

India has a large variation in climate from region to region due to its vast size. The climate ranges from extremely hot desert regions to high altitude locations with severely cold conditions. Climate zones are classified into 5 distinct types: cold, composite, hot and dry, hot and humid, and moderate. Select the checkbox corresponding to the Climate Zone layer to view its mapping and the corresponding Legend popup as shown in Figure 5-4.

FDSS



Fire Decision Support System



Figure 5-4: Map showing climatic zones in India

Note:

To view wind, seismic or climatic zone information at a particular location in India, use the Identify a feature tool as described in Section 3.6

5.2 Country layer

This section describes the various sub-layers under the Country layer. The tree view of the Country layer is shown in Figure 5-5 below.





5.2.1 Fire Station Density in India

FDSS has classified the density of fire stations in India into 6 different categories. Select the Fire Stations in India checkbox to view its mapping and the corresponding Legend popup as shown in Figure 5-6.




Figure 5-6: Fire Station Density in India

5.2.2 Operational Fire Stations

Operational Fire Stations layer include all working fire stations in the country and which are available for fire emergency. Select the Operational Fire Stations checkbox to view the operational fire stations in the Map window as shown in Figure 5-7.





Figure 5-7: Operational Fire Stations in India

5.2.3 Proposed Fire Stations

Proposed Fire Stations include all the fire stations which are proposed in addition to Operational Fire stations. These proposed fire stations are derived from response time based GIS analysis based on road networks and risk categories. Select the Proposed Fire Stations checkbox to view the proposed rural and urban fire stations in the Map window as shown in Figure 5-8.





Figure 5-8: Proposed Fire Stations in India

5.2.4 Other Agency Fire Stations Area

Other agency fire station area includes service area being served by other agencies such as private establishments in industrial estates, fire stations under thermal power plants, public sector industrial compounds etc. Select the Other Agency Fire Station Area checkbox to view the layer in the Map window.

The user has to zoom into a particular area in the Map window, to be able to view this layer. As an example, Figure 5-9 shows one such area served by other agency fire station in Ahmadabad, Gujarat.

FDSS







Figure 5-9: Other Agency Fire Stations Area

5.2.5 District Fire Risk Category

The project team has categorized districts of India into 5 fire risk categories (viz. Very High, High, Medium, and Low) based on weighted overlay of various hazard and risk factors. Select the District Fire Risk Category checkbox to view the district level risk and the corresponding Legend popup as shown in Figure 5-10. For more details refer to Section 4.2





Figure 5-10: District Fire Risk Category in India

Note: To view fire risk category information at a particular location in India, use the Identify a feature 100 tool as described in Section 3.6.

5.3 States and Union Territories layer

As mentioned earlier in Section 3.3.2.4, The States and Union Territories layer includes thirty five layers, each devoted to a single state or union territory. Under each State or Union Territory layer, there are several sub layers, namely, Major Roads, Minor Roads, Rail Network, Land Use Land Cover, Built up Area and Ideal urban fire station jurisdiction.

As an example, the expanded tree view of Delhi State illustrating the sub-layers is shown in Figure 5-11. However, not all the States or Union Territories would necessarily have all the sub layers.





Figure 5-11: Expanded view of Delhi state illustrating sub-layers

5.3.1 Major Roads

Select the Major Roads checkbox to view it in the Map window as shown in Figure 5-12. The Major roads include highways and main roads for each State/ UT.



Figure 5-12: Major Roads of Delhi



5.3.2 Minor Roads

Select the Minor Roads checkbox to view it in the Map window as shown in Figure 5-13. Minor Roads include some Bypass roads, Streets and Other minor roads.



Figure 5-13: Minor Roads of Delhi



5.3.3 Rail Network

Select the Rail Network checkbox to view it in the Map window as shown in Figure 5-14.



Figure 5-14: Rail Network of Delhi



5.3.4 Land Use Land Cover

Land Use Land Cover involves the classification of land into various categories by classifying raw satellite data. Basically, the Land Use and Land Cover (LULC) data files describe the vegetation, water bodies, natural surface, and cultural features on the land surface. Select the checkbox corresponding to land use land cover to view its mapping as shown in Figure 5-15.



Figure 5-15: Land Use Land Cover of Delhi



5.3.5 Built-up Area

Built up area represents the classified building footprints in an area according to its usage and the types of structures. Select the Built up Area checkbox to view its mapping and the corresponding Legend popup as shown in Figure 5-16.



Figure 5-16: Built-up area of Delhi



5.3.6 Ideal Urban Fire Station Jurisdiction

This layer shows the ideal service area of an operational urban fire station based on response time of a firefighting vehicle to reach the fire scene. Ideal Urban fire station jurisdiction areas are delineated for redeploying proper coverage of each fire station through GIS analysis. This analysis comes from travel time taken by a fire engine to reach the place of fire incidence and risk assessment in the area. Select the Ideal Urban Fire Station Jurisdiction checkbox to view its mapping as shown in Figure 5-17.



Figure 5-17: Ideal Urban Fire Station Jurisdictions of Delhi



5.4 Administrative Boundaries layer

The Administrative Boundaries layer includes three layers, namely, Phase-Wise States, State Capital and Districts layers. The expanded tree view of the Administrative Boundaries layer illustrates the complete menu and is shown in Figure 5-18.





5.4.1 Phase-Wise States

This layer displays the states covered under the five different phases of the development of the Fire Hazard and Risk Analysis project. These States can be viewed by selecting the Phase-Wise States checkbox under Administrative Boundaries layer as shown by the red highlighted rectangle in Figure 5-19.



Figure 5-19: Map showing phase wise states in India





5.4.2 State/UT Capital

This layer displays the state capitals of all the Indian states/UTs. The state/UT capitals can be viewed by selecting the State/UT Capital checkbox under Administrative Boundaries layer as shown in Figure 5-20.



Figure 5-20: Map showing State Capital



5.4.3 Districts

This layer displays all the districts in India along with their boundaries. The district boundaries can be viewed by selecting the Districts checkbox under Administrative Boundaries as shown by the red highlighted rectangle in Figure 5-21.



Figure 5-21: Map showing districts in India



6 User Management

As already stated in the Getting Started section, FDSS has two types of users, namely:

- 1. Administrator
- 2. Normal User

The User button ^(a) on the toolbar is enabled only when a user logs in as an Administrator (Refer to area highlighted by red rectangle in the toolbar in Figure 6-1).

1. Click on the User button <a>In the toolbar to display the User Management window. This window lists all existing Administrative and Normal users in the FDSS as shown in Figure 6-1 below with the red highlighted rectangle at the centre.



Figure 6-1: User Management window

The user management window is depicted in Figure 6-2 to illustrate the various management tasks which can be carried out by an FDSS administrator. The User (Orange Circle in Figure 6-2) column lists the Usernames and Group (Red Circle in Figure 6-2) column lists the corresponding User Type.



🛞 pieronline.rmsi	.com/FDSS/UserMgr	mt.aspx			
			User	Group	
Change Password	Change Group R	emove	RMSI	Administrator	
Change Password	Change Group Re	emove	user2	Administrator	
Change Password	Change Group Re	emove	user3	Normal User	
Change Password	Change Group Re	emove	user4	Administrator	
Change Password	Change Group Re	emove	user5	Administrator	
Change Password	Change Group Re	emove	user6	Administrator	

Figure 6-2: User Management window illustrating Administrator tasks

6.1 Change password

1. To change the password, click the Change Password link in the first column (Red Box in Figure 6-2) corresponding to the user whose password needs to be changed. This displays the Change Password window as shown in Figure 6-3.

🕙 Change Passwor	d - Mozilla Firefox	
🕘 pieronline.rmsi.com	/FDSS/ChangePasswd.aspx?User=user3	₹2
Username:	user3	
Old Password:	•••••	
New Password:	•••••	
Confirm Password:	••••••	
	OK Cancel	

Figure 6-3: Change Password window

- Fill in the old password in Old Password textbox, new password in New Password textbox and retype new password in Confirm Password textbox as shown in Figure 6-3.
- 3. Click the OK button to change the password. The new password gets activated.
- 4. Click Cancel to abort the operation and return to the previous screen.



Note:

- Make sure the old password entered is correct. No change is done if the old password is incorrect.
- Only alphanumeric characters are allowed.

6.2 Change Group

1. To change the group, click the Change Group link in the second column (Orange Box in Figure 6-2) corresponding to the user whose group needs to be changed. This displays the Change Group window as shown in Figure 6-4.

Username: user2 Group: Administrator Administrator Normal User UK Cancel	-
Administrator Normal User	
Normal User	
UK	

Figure 6-4: Change Group window

- 2. Select Administrator or Normal User from the Group dropdown list.
- 3. Click OK to change the group.
- 4. Click Cancel to abort the operation and return to the previous screen.

6.3 Add New User

1. To add a new user click the New User button at the bottom left of the User Management window (Purple box in Figure 6-2). This displays the Create New User window as shown in Figure 6-5.



🅘 Create N	ew User - Mozilla Firefox	W.McAfee	• • • • ×
🛞 pieronli	ne. rmsi.com /FDSS/CreateUser.aspx		合
Username: Group: Password:	User7 Administrator		

Figure 6-5: Create New User window

- 2. Type a user name in the Username textbox.
- 3. Select the Group from the Group dropdown. The two options available are Administrator and Normal User.
- 4. Type in a password in the Password textbox.
- 5. Click OK to add a new user to the system.
- 6. Click Cancel to abort the operation and return to the previous screen.

6.4 Remove an existing user

- 1. Click the Remove link as shown in the User Management window (Green Box in Figure 6-2) to remove the user from the list.
- 2. A prompt pop-up opens on the screen to confirm/cancel the operation as shown in Figure 6-6.

🕘 User Management - Mozilla	Firefox W.McAfee	
🛞 pieronline.rmsi.com/FDSS/	UserMgmt.aspx	合
Change Password Change C		*
Change Password Change		
Change Password Change C	Are you sure you want to delete this user?	
Change Password Change (-	E
Change Password Change		
Change Password Change (OK Cancel	
Change Password Change C		
Change Password Change G	roup Remove Chandra Shekhar Administrator	*

Figure 6-6: Remove user prompt window



- 3. Click OK to delete the selected user. The deleted user will not be able to log into the system with the earlier assigned user name and password.
- 4. Click Cancel to abort the operation and return to the previous screen.



7 Annexure: FDSS desktop application

Fire Decision Support System has been developed as an exposure and risk tracking solution that utilizes geospatial technologies to improve identification, assessment and monitoring of fire risk across the country. Along with the FDSS web application (available to all authorized users); an FDSS desktop application has also been developed (available only to users with administrative rights). The FDSS desktop application facilitates entering large amounts of data on fire stations in a quick, easy, and efficient manner. The desktop application provides two tools viz., Data Import Utility tool and Master Table editing tool :

- 1. Data Import Utility tool contains data on all existing fire stations. It allows administrative users to add new fire stations and edit (add, alter, delete) data on existing fire stations.
- 2. Master Table editing tool allows administrative users to edit (add, alter, delete) the master table containing data on all the ideal and new proposed fire stations in India that have been generated by applying GIS analysis as part of this project.

This Annexure provides users step-by-step assistance in using the Data Import Utility tool and the Fire DSS Editing tool effectively.

7.1 Data Import Utility tool

7.1.1 Opening the tool

Double click on the Data Import Utility tool icon **(on the desktop of Administrator user** machines) to start the application. The tool initiates as shown in Figure 7-1.



						and Nicobar Islands	V District :	Al	× .	Search Show	AL
10										_	
FireStationId	Reference No.	Fire Station Name	State	District	Municipality	Town/City	Latitude	Longitude	Phone No.	SurveyedBy	SurveyDate
410	RJ410	Banipark Fire Sta	Rajasthan	Jaipur	Jaipur	Jaipur	26.931667	75.785500	01412201898	Rohit Singh	9/3/2011
412	RJ412	Ghat Gate Fire St	Rajasthan	Jaipur	Jaipur	Jaipur	26.913167	75.832500	01412615550	Rohit Singh	9/3/2011
414	RJ414	22 Godam Fire St	Rajasthan	Jaipur	Jaipur	Jaipur	26.896167	75.789000	01412211258	Rohit Singh	9/3/2011
418	MH418	Adharwadi Fire a	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.248833	73.016833	02512310155	Sunny Gupta	8/23/2011
419	RJ419	Vishwa Karma In	Rajasthan	Jaipur	Jaipur	Jaipur	26.973167	75.777167	01412332573	Rohit Singh	9/3/2011
420	MH420	MIDC Fire Station	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.211500	73.109500	02512470357	Sunny Gupta	8/23/2011
422	RJ422	Sitapura Fire Stati	Rajasthan	Jaipur	Jaipur	Jaipur	26.784667	75.843000	01412175250	Rohit Singh	9/4/2011
423	RJ423	Amer Fire Station	Rajasthan	Jaipur	Jaipur	Amer	26.992333	75.874333	01412531282	Rohit Singh	9/5/2011
426	RJ426	Azad Park Fire St	Rajasthan	Ajmer	Ajmer	Ajmer	26.469167	74.640000	01452429000	Rohit Singh	9/6/2011
427	JK427	Shaheedi Chowk	Jammu and Kash	Jammu	Jammu	Jammu	32.712167	74.871000		Sushil Gupta, Uji	8/20/2011
428	RJ428	Pushkar Fire Stati	Rajasthan	Ajmer	Pushkar	Pushkar	26.492500	74.558667	014522772513	Rohit Singh	10/5/2011
429	JK429	Gangyal	Jammu and Kash	Jammu	Jammu	Jammu	32.671667	74.868500	01912480026	Ujjwal, Abhinav	8/20/2011
430	MH430	Pimpri Main Fire	Maharashtra	Pune	Pimpri Chinchwad	Pimpri	18.622500	73.818167	02027423333	Paramjeet Dalal	8/26/2011
435	MH435	Bhosari Fire Station	Maharashtra	Pune	Pimpri Chinchwad	Bhosari	18.619833	73.845333	02027120090	Paramjeet Dalal	8/26/2011
436	JK436	New Secretariat	Jammu and Kash	Jammu	Jammu	Jammu	32.735167	74.860500		Sushil Gupta	8/20/2011
441	JK441	Mubarak Mandi F	Jammu and Kash	Jammu	Jammu	Jammu	32.738833	74.875333	01912570036	Sushil Gupta	8/20/2011
442	MH442	Rahatini Fire Stati	Maharashtra	Pune	Pimpri Chinchwad	Rahatini	18.596333	73.781833	02027276749	Paramjeet Dalal	8/26/2011
443	MH443	Pradhikaran Fire	Maharashtra	Pune	Pimpri Chinchwad	Pradhikaran	18.659333	73.771833	02027652066	Paramjeet Dalal	8/26/2011
444	MH444	Hadaspar Fire St	Maharashtra	Pune	PMC	Pune	18.512500	73.915333	02026870207	Paramjeet Dalal	8/28/2011
446	RJ446	Jaipur City Civil D	Rajasthan	Jaipur	Jaipur	Jaipur	26.917667	75.803000	01412375925	Rohit Singh	9/4/2011
448	RJ448	Jaipur Collectrate	Rajasthan	Jaipur	Jaipur	Jaipur	26.917833	75.802667	01412375925	Rohit Singh	9/4/2011
449	MH449	Lt. Dayaram Rajg	Maharashtra	Pune	PMC	Pune	18.529667	73.870500	02026059230	Paramjeet Dalal	8/27/2011
457	MH457	Pune Link Road	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.224500	73.139000	02512365101	Sunny Gupta	8/23/2011
458	MH458	Yerwada Fire Sta	Maharashtra	Pune	PMC	Pune	18.550167	73.879167	02026696400	Paramjeet Dalal	8/27/2011
459	RJ459	Ajmer Civil Defen	Rajasthan	Ajmer	Ajmer Municipal	Ajmer	26.469167	74.641167	01452623779	Rohit Singh	9/6/2011
460	MH460	Aundh Fire Station	Maharashtra	Pune	PMC	Pune	18.560833	73.814333	02025851788	Paramjeet Dalal	8/27/2011
461	MH461	Kasbapeth Fire S	Maharashtra	Pune	PMC	Pune	18.522167	73.856500	02024578950	Paramjeet Dalal	8/28/2011

Figure 7-1: Data Import Utility tool UI

7.1.2 Data Import Utility tool interface

The interface has 4 parts as shown in Figure 7-2.

- 1. Menu bar (red rectangle highlight)
- 2. Tool bar (blue rectangle highlight)
- 3. Status bar
- 4. Document window (green rectangle highlight)

FDSS





Import Utili	ty of Fire Statio	ns									l
		Menu	bar								
- 1	'ool bar										
				2.24	Search Option						
					State : Andaman	and Nicobar Islands	V District :	All	× _	Search Show	All
		Sea	arch Option	frame							
									Document v	vindow	
	_								Document	vindow	
1	Data fram	le									
			_								
ireStationId	Reference No.	Fire Station Name	State	District	Municipality	Town/City	Latitude	Longitude	Phone No.	SurveyedBy	SurveyDate
10	RJ410	Banipark Fire Sta	Rajasthan	Jaipur	Jaipur	Jaipur	26.931667	75.785500	01412201898	Rohit Singh	9/3/2011
12	RJ412	Ghat Gate Fire St	Rajasthan	Jaipur	Jaipur	Jaipur	26.913167	75.832500	01412615550	Rohit Singh	9/3/2011
4	RJ414	22 Godam Fire St	Rajasthan	Jaipur	Jaipur	Jaipur	26.896167	75.789000	01412211258	Rohit Singh	9/3/2011
8	MH418	Adharwadi Fire a	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.248833	73.016833	02512310155	Sunny Gupta	8/23/2011
19	RJ419	Vishwa Karma In	Rajasthan	Jaipur	Jaipur	Jaipur	26.973167	75.777167	01412332573	Rohit Singh	9/3/2011
20	MH420	MIDC Fire Station	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.211500	73.109500	02512470357	Sunny Gupta	8/23/2011
22	RJ422	Sitapura Fire Stati	Rajasthan	Jaipur	Jaipur	Jaipur	26.784667	75.843000	01412175250	Rohit Singh	9/4/2011
23	RJ423	Amer Fire Station	Rajasthan	Jaipur	Jaipur	Amer	26.992333	75.874333	01412531282	Rohit Singh	9/5/2011
26	RJ426	Azad Park Fire St	Rajasthan	Ajmer	Ajmer	Ajmer	26.469167	74.640000	01452429000	Rohit Singh	9/6/2011
27	JK427	Shaheedi Chowk	Jammu and Kash	Jammu	Jammu	Jammu	32.712167	74.871000		Sushil Gupta, Ujj	8/20/2011
28	RJ428	Pushkar Fire Stati	Rajasthan	Ajmer	Pushkar	Pushkar	26.492500	74.558667	014522772513	Rohit Singh	10/5/2011
29	JK429	Gangyal	Jammu and Kash	Jammu	Jammu	Jammu	32.671667	74.868500	01912480026	Ujjwal, Abhinav	8/20/2011
30	MH430	Pimpri Main Fire	Maharashtra	Pune	Pimpri Chinchwad	Pimpri	18.622500	73.818167	02027423333	Paramjeet Dalal	8/26/2011
35	MH435	Bhosari Fire Station	Maharashtra	Pune	Pimpri Chinchwad	Bhosari	18.619833	73.845333	02027120090	Paramjeet Dalal	8/26/2011
36	JK436	New Secretariat	Jammu and Kash	Jammu	Jammu	Jammu	32.735167	74.860500		Sushil Gupta	8/20/2011
11	JK441	Mubarak Mandi F	Jammu and Kash	Jammu	Jammu	Jammu	32.738833	74.875333	01912570036	Sushil Gupta	8/20/2011
12	MH442	Rahatini Fire Stati	Maharashtra	Pune	Pimpri Chinchwad	Rahatini	18.596333	73.781833	02027276749	Paramjeet Dalal	8/26/2011
13	MH443	Pradhikaran Fire	Maharashtra	Pune	Pimpri Chinchwad	Pradhikaran	18.659333	73.771833	02027652066	Paramjeet Dalal	8/26/2011
14	MH444	Hadaspar Fire St	Maharashtra	Pune	PMC	Pune	18.512500	73.915333	02026870207	Paramjeet Dalal	8/28/2011
16	RJ446	Jaipur City Civil D	Rajasthan	Jaipur	Jaipur	Jaipur	26.917667	75.803000	01412375925	Rohit Singh	9/4/2011
18	RJ448	Jaipur Collectrate	Rajasthan	Jaipur	Jaipur	Jaipur	26.917833	75.802667	01412375925	Rohit Singh	9/4/2011
19	MH449	Lt. Dayaram Rajg	Maharashtra	Pune	PMC	Pune	18.529667	73.870500	02026059230	Paramjeet Dalal	8/27/2011
57	MH457	Pune Link Road	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.224500	73.139000	02512365101	Sunny Gupta	8/23/2011
58	MH458	Yerwada Fire Sta	Maharashtra	Pune	PMC	Pune	18.550167	73.879167	02026696400	Paramjeet Dalal	8/27/2011
59	RJ459	Ajmer Civil Defen	Rajasthan	Ajmer	Ajmer Municipal	Ajmer	26.469167	74.641167	01452623779	Rohit Singh	9/6/2011
50	MH460	Aundh Fire Station	Maharashtra	Pune	PMC	Pune	18.560833	73.814333	02025851788	Paramjeet Dalal	8/27/2011
51	MH461	Kasbapeth Fire S	Maharashtra	Pune	PMC	Pune	18.522167	73.856500	02024578950	Paramjeet Dalal	8/28/2011
		Cinched Deed Fit		1		0	10.475500	73.015033	0000405150	Deverient Datel	0/20/2011

Figure 7-2: Window highlighting Data Import Utility tool interface

7.1.2.1 Menu bar: File menu

The Data Import Utility tool comprises of one menu element i.e. File menu. Clicking the File menu displays two sub menus, namely, Add New Fire Station and Exit. Figure 7-3 displays the File menu and its sub menus.

File		
	Add New Fire Station	Ctrl+N
	Exit	

Figure 7-3: File menu with its sub menu

7.1.2.1.1 Add a new fire station

The main purpose of the Data Import Utility tool is to add data on new fire stations. The following steps show how to use the Add New Fire Station functionality.



1. Click the File menu in the menu bar and then click Add New Fire Station⁸ as shown in Figure 7-4 (highlighted by red rectangle).

		1									
dd New Fire Sta	tion Ctrl+N										
xit					- Search Option						
		9			State : Andaman	and Nicobar Islands	V District :	All	v .	Search Show	All
	Reference	Fire Station	C 1.1	District	M	T (C)	1.0.1			c	0.01
FireStationId	No.	Name	State	District	Municipality	Town/City	Latitude	Longitude	Phone No.	SurveyedBy	SurveyDate
10	RJ410	Banipark Fire Sta	Rajasthan	Jaipur	Jaipur	Jaipur	26.931667	75.785500	01412201898	Rohit Singh	9/3/2011
112	RJ412	Ghat Gate Fire St	Rajasthan	Jaipur	Jaipur	Jaipur	26.913167	75.832500	01412615550	Rohit Singh	9/3/2011
114	RJ414	22 Godam Fire St	Rajasthan	Jaipur	Jaipur	Jaipur	26.896167	75.789000	01412211258	Rohit Singh	9/3/2011
18	MH418	Adharwadi Fire a	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.248833	73.016833	02512310155	Sunny Gupta	8/23/2011
19	RJ419	Vishwa Karma In	Rajasthan	Jaipur	Jaipur	Jaipur	26.973167	75.777167	01412332573	Rohit Singh	9/3/2011
20	MH420	MIDC Fire Station	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.211500	73.109500	02512470357	Sunny Gupta	8/23/2011
-22	RJ422	Sitapura Fire Stati	Rajasthan	Jaipur	Jaipur	Jaipur	26.784667	75.843000	01412175250	Rohit Singh	9/4/2011
23	RJ423	Amer Fire Station	Rajasthan	Jaipur	Jaipur	Amer	26.992333	75.874333	01412531282	Rohit Singh	9/5/2011
26	RJ426	Azad Park Fire St	Rajasthan	Ajmer	Ajmer	Ajmer	26.469167	74.640000	01452429000	Rohit Singh	9/6/2011
27	JK427	Shaheedi Chowk	Jammu and Kash	Jammu	Jammu	Jammu	32.712167	74.871000		Sushil Gupta, Ujj	8/20/2011
28	RJ428	Pushkar Fire Stati	Rajasthan	Ajmer	Pushkar	Pushkar	26.492500	74.558667	014522772513	Rohit Singh	10/5/2011
29	JK429	Gangyal	Jammu and Kash	Jammu	Jammu	Jammu	32.671667	74.868500	01912480026	Ujjwal, Abhinav	8/20/2011
30	MH430	Pimpri Main Fire	Maharashtra	Pune	Pimpri Chinchwad	Pimpri	18.622500	73.818167	02027423333	Paramjeet Dalal	8/26/2011
35	MH435	Bhosari Fire Station	Maharashtra	Pune	Pimpri Chinchwad	Bhosari	18.619833	73.845333	02027120090	Paramjeet Dalal	8/26/2011
36	JK436	New Secretariat	Jammu and Kash	Jammu	Jammu	Jammu	32.735167	74.860500		Sushil Gupta	8/20/2011
41	JK441	Mubarak Mandi F	Jammu and Kash	Jammu	Jammu	Jammu	32.738833	74.875333	01912570036	Sushil Gupta	8/20/2011
42	MH442	Rahatini Fire Stati	Maharashtra	Pune	Pimpri Chinchwad	Rahatini	18.596333	73.781833	02027276749	Paramjeet Dalal	8/26/2011
43	MH443	Pradhikaran Fire	Maharashtra	Pune	Pimpri Chinchwad	Pradhikaran	18.659333	73.771833	02027652066	Paramjeet Dalal	8/26/2011
44	MH444	Hadaspar Fire St	Maharashtra	Pune	PMC	Pune	18.512500	73.915333	02026870207	Paramjeet Dalal	8/28/2011
46	RJ446	Jaipur City Civil D	Rajasthan	Jaipur	Jaipur	Jaipur	26.917667	75.803000	01412375925	Rohit Singh	9/4/2011
48	RJ448	Jaipur Collectrate	Rajasthan	Jaipur	Jaipur	Jaipur	26.917833	75.802667	01412375925	Rohit Singh	9/4/2011
49	MH449	Lt. Dayaram Rajg	Maharashtra	Pune	PMC	Pune	18.529667	73.870500	02026059230	Paramjeet Dalal	8/27/2011
157	MH457	Pune Link Road	Maharashtra	Thane	Kalyan Dombivali	Kalyan Dombivali	19.224500	73.139000	02512365101	Sunny Gupta	8/23/2011
158	MH458	Yerwada Fire Sta	Maharashtra	Pune	PMC	Pune	18.550167	73.879167	02026696400	Paramjeet Dalal	8/27/2011
159	RJ459	Ajmer Civil Defen	Rajasthan	Ajmer	Ajmer Municipal	Ajmer	26.469167	74.641167	01452623779	Rohit Singh	9/6/2011
	MH460	Aundh Fire Station	Maharashtra	Pune	PMC	Pune	18.560833	73.814333	02025851788	Paramjeet Dalal	8/27/2011
60		the second second second									
160 161	MH461	Kasbapeth Fire S	Maharashtra	Pune	PMC	Pune	18.522167	73.856500	02024578950	Paramjeet Dalal	8/28/2011

Figure 7-4: Window showing File menu to add new fire station (highlighted by red rectangle)

2. A form is displayed as shown in Figure 7-5. This form comprises of 8 pages, with each page allowing the user to enter different types of information on the new fire station.

⁸ To add a new fire station user can also click on in the toolbar. It will open the same form as shown in Figure 7-2 to enter fire station details.

FDSS



Fire Decision Support System

🖶 Fire Station	General Information				
			FS R	ef#: AN4	4032
Geography					
State :	Andaman and Nicobar Islands 💌	City/Town :			
District :	Nicobar 🔽		Degree	Minute	Second
Municipality :		Lat :	0.0	0.0	0.0
		Lon :	0.0	0.0	0.0
General Inform	ation				
Name :					
Address :					
Phone No. : 1)			2)		
Fax No. :		Emergency N	o. :		
FS Type :	Urban 🗸				
Officer in-charg	ge Details				
Name :					
Designation :	×	Mobile No	p. :		
Administrative	Details				
Admin Depa	irtment :	*			
Officer	Name:				
Desig	nation :	~			
Mob	ile No. :				
Address/lo	ocation :				
No. of fire :	station falling under above administra	tion/ jurisdiction	: [(
Surveyed By :		S	urvey Date :	3/ 1/201	3 💌
			Next	>> (Cancel
Page 1 of 8					

Figure 7-5: Form showing Fire Station General Information page

3. Fill in the details in the form⁹. A sample form showing the last page is shown in Figure 7-6.

⁹ A form comprise of various controls such as text boxes, drop down etc. Make sure that correct state and district is selected in the form and other details are filled correct as well.

FDSS



Fire Decision Support System

Month	Total Calls(A+B+C+D)	Total Fire Incidence calls(A)	Residential	Industrial	Institutional/comme	Others(fire Incidence)	Total Rescue Incidence(B)	Road Accidents	Building collapse	
Add b	uttop									
	utton	101								>
+	Delete	In such that we are								
<u>+</u>	Delete	button								
+	ity of fire incidences						_			
+		Dutton Total no. of Medium Fire Incidence	Total No. of Serious fire Incidence	Brief description of N	Major Fire Incidence]			
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					
	ity of fire incidences Total no. of small Fire	Total no. of Medium Fire	Serious fire	Brief description of N	Major Fire Incidence					

Figure 7-6: Form showing Fire Calls and Incidence Details page

4. Click on the Save button¹⁰ to save the newly entered details as shown in Figure 7-6 (highlighted by red rectangle). This displays a message box as shown in Figure 7-7.

Fire stati	on added successfully
C	ОК

Figure 7-7: Message box displayed on successful addition of a fire station

5. Click on OK. This displays a dialog box (Figure 7-8) prompting on whether user wishes to update ideal fire station details for this fire station.

¹⁰ Save button is given on the last (eighth) page.





Figure 7-8: Dialog box showing question to update ideal fire station details

6. Click on Yes button, to choose the corresponding new fire station proposed by RMSI team. This is because, the new fire station location/name proposed by RMSI team is to suggest approximate location to fire service for construction of new fire station. However, in reality, the selection of location depends on many factors like availability of land, optimum location, its cost etc. Now through this tool, we try to correlate the newly operational fire station with the corresponding proposed one and perform the Infrastructural Gap analysis. Refer to section 7.1.2.1.2 to get overview of this process. To view a new created fire station in a data frame refer to steps mentioned in section 7.1.2.4.1.

Click on No button if the location/ name of newly operational fire station is different from that recommended by RMSI team in the corresponding State/UT report. In that case, the user needs to perform separate response time analysis for this fire station and manually enter the ideal fire station details in the Master Table using Master Table Editing Tool.

7.1.2.1.2 Moving a new fire station to ideal fire station table

As discussed above (section 7.1.2.1.1), when a new fire station is added in the database, user has to select corresponding proposed fire station location from the proposed list of fire stations and move it to the ideal fire station table. To perform this task, follow the steps mentioned below:

- 1. Repeat step 1 to step 5 as mentioned in section 7.1.2.1.1.
- Click on Yes button (refer section 7.1.2.1.1, point 6). A dialog box containing details of new fire stations proposed by RMSI team would be displayed. This list includes the proposed fire stations located within 30 km radius of the newly added fire station (Figure 7-9¹¹). A dialog box would not display any fire station if they are not within 30 km radius of the newly added fire station.

¹¹ Calculation is based on the latitude and longitude specified while entering new fire station details.



 Proposed Fire Static		T		1		1	
District	DistrictCode	FireStationId	FSRefNo	Name	TOWNCITY	OparationalType	Area
 West Tripura	1183		TR_New_Urban	New Urban		New Urban	6.09
West Tripura	1183		TR_New_Urban	New Urban		New Urban	11.07
West Tripura	1183		TR_New_Rural	Takarjala Fir		New Rural	249.5
West Tripura	1183		TR_New_Rural	Rabigopal P		New Rural	221.5
West Tripura	1183		TR_New_Rural	Usha Bazaa		New Rural	43.14
West Tripura	1183		TR_New_Rural	Sekerkote F		New Rural	118.5
West Tripura	1183		TR_New_Rural	Ananda Na		New Rural	75.18
West Tripura	1183		TR_New_Rural	Kali Bazaar		New Rural	66.42

Figure 7-9: Window showing list of all the proposed fire stations within 30 km radius of the new fire station

- 3. Select the row corresponding to the proposed fire station for which ideal fire station details need to be updated as shown in Figure 7-9.
- 4. Click on the Update button as shown in Figure 7-9 to apply the changes.
- 5. The changes made will be reflected in both the existing fire station table and ideal fire station table.

7.1.2.2 Tool bar

Data Import Utility tool provides the Add a New Fire Station tool \square in the tool bar. Click this tool to add a new fire station as explained in section 7.1.2.1.1. This tool and the path run though File menu to add a new fire station perform a same functionality.

7.1.2.3 Status bar

The Status bar in the Data Import Utility tool displays the total number of records corresponding to the selected fire station.

7.1.2.4 Document window

The Document window (Figure 7-2) is divided into two frames as follows:

- 1. Search Option frame: provides the list of existing fire stations based on the state and district selected from the State and District dropdown respectively.
- 2. Data frame: displays the list of all existing fire stations. Initially when the application initiates, all the fire stations of India are displayed in a Data frame. User can limit the data displayed by using the Search option. It also provides the option to add or delete a fire station.

Working with Search Option and Data frames is described in detail in section 7.1.2.4.1.



7.1.2.4.1 Document window: Searching fire stations

Data Import Utility allows the user to search for fire stations in the following way:

1. Select a state from the State dropdown. As an example, Delhi state is selected (Highlighted by red rectangle in Figure 7-10).

					Search Option		V District :	A.I.		Search Show	
					State : Delhi Delhi		 District : 	All	ĭ	Search Show	
					Goa Gujarat						
					Haryan	a					
					Jammu	al Pradesh and Kashmir					
					Jharkha Karnata		~				
FireStationId	Reference	Fire Station	State	District	Municipality	Town/City	Latitude	Longitude	Phone No.	SurveyedBy	SurveyDate
956	No. DL956	Name Mayur Place	Delhi	East	MCD	Delhi	28.591000	77.299167		Rakesh, Ujiwal,	8/2/2011
960	DL356	Mayur Vihar	Delhi	East	MCD	Delhi	28.625167	77.307333	0112272110	Rakesh, Ujjwal,	8/2/2011
961	DL961	Tahirpur	Delhi	North East	MCD	Delhi	28.685000	77.315167	0112272110	Rakesh, Ujiwal,	8/2/2011
962	DL962	Shashtri Park	Delhi	North East	MCD	Delhi	28.675500	77.263000	01122853933	Rakesh, Ujiwal a	8/2/2011
963	DL963	Gokulpur	Delhi	North East	MCD	Delhi	28.700500	77.291333	01122815612	Rakesh, Ujjwal a	8/2/2011
964	DL964	Delhi Secretariat	Delhi	New Delhi	MCD	Delhi	28.629000	77.251833	01123392677	Rakesh, Ujjwal,	8/3/2011
965	DL965	Shahdara	Delhi	East	MCD	Delhi	28.674167	77.284167	01122323322	Rakesh, Ujjwal,	8/2/2011
966	DL966	Geeta Colony	Delhi	East	MCD	Delhi	28.653500	77.270167	01122513330	Rakesh, Ujjwal A	8/2/2011
967	DL967	Rohini Sector 5	Delhi	North West	MCD	Delhi	28.718500	77.109833	01127042777	Anand, Abhinav	8/5/2011
968	DL968	Okhla Phase III	Delhi	South	MCD	Delhi	28.552167	77.269500	01126847511	Abhinav, Anand,	8/9/2011
969	DL969	Wazirpur	Delhi	North West	MCD	Delhi	28.701167	77.164833	01127372621	Anand and Abhin	8/5/2011
970	DL970	Keshavpuram	Delhi	West	MCD	Delhi	28.681333	77.147500	01127189090	Anand, Abhinav	8/2/2011
971	DL971	Najafgarh	Delhi	South West	MCD	Delhi	28.612000	76.976667	01128012214	Abhinay, Anand,	8/9/2011
972	DL972	Mangolpuri	Delhi	West	MCD	Delhi	28.686167	77.102000	01127015280	Anand, Abhinav	8/2/2011
973	DL973	Sarita Vihar	Delhi	South	MCD	Delhi	28.532333	77.296667	01129945066	Abhinav, Anand	8/9/2011
974	DL974	Okhla Phase I	Delhi	South	MCD	Delhi	28.523500	77.282667	01126371344	Abhinav, Anand	8/9/2011
975	DL975	Dwarka Sector 6	Delhi	South West	MCD	Delhi	28.597333	77.064167	01128082565	Anand, Ujjwal	8/10/2011
976	DL976	Narela	Delhi	North West	MCD	Delhi	28.848833	77.087833	01127785311	Abhinav, Anand	8/4/2011
978	DL978	Rashtrapati Bha	Delhi	Central	NDMC	Delhi	28.618833	77.197333	01123015321	Sushil Gupta and	9/14/2011
979	DL979	Harinagar	Delhi	West	MCD	Delhi	28.622333	77.110333	01125141433	Anand, Ujiwal	8/10/2011
980	DL980	Mathura Road	Delhi	South	MCD	Delhi	28.572500	77.257667	01126341301	Abhinav, Ujjwal a	8/9/2011
981	DL981	Janakpuri	Delhi	West	MCD	Delhi	28.621333	77.091000	01125522000	Anand, Ujjwal	8/10/2011
982	DL982	Nehru Place	Delhi	South	MCD	Delhi	28.549000	77.256333	01126445230	Abhinav, Anand	8/9/2011
983	DL983	Jahangir Puri	Delhi	North West	MCD	Delhi	28.733833	77.174500	01127636006	Abhinav, Anand,	8/4/2011
984	DL984	Bhikaji Cama Place	Delhi	South West	MCD	Delhi	28.568500	77.183833	01126173583	Abhinav, Anand	8/9/2011
985	DL985	Rohini Sector 16	Delhi	North West	MCD	Delhi	28.731000	77.121000	01127571194	Anand, Ujjwal, B	8/5/2011
Cert Mark	DL986	Bhorgarh	Delhi	North West	MCD	Delhi	28.821500	77.102000	01127784996	Bhanu, Abhinav	8/4/2011
986	DL306	Briolgan	Dem	HOIGI WOOL	MCD	Donn	20.021000	11.102000		briding, rabilitar	or the born

Figure 7-10: Window showing existing fire station details of Delhi state

- 2. The user can select one of two options in the District dropdown:
 - a) **Select All option from District dropdown:** This option generates a list of all the fire stations within a selected State/UT. As an example 'All' option is selected. (highlighted by blue rectangle in Figure 7-10)
 - b) **Select a particular district from District dropdown:** This option generates a list of all the fire stations within the selected district and state.
- 3. Click the Search button as shown in Figure 7-10 (orange rectangle highlight) to generate a list of existing fire stations based on the search criteria selected in step 1 and step 2. In Figure 7-10, details of all the existing fire stations in Delhi are populated in Data frame.(highlighted by green rectangle)
- 4. Click the Show All button (to the right of the Search button) to view a list of all the fire stations of India.



7.1.2.4.2 Document Window: Modifying fire station details

Master Table editing tool provides the user with a facility to edit (modify, add, delete) information on an existing fire station. This function is useful to correct wrong data added earlier or to update existing data with changed data.

7.1.2.4.2.1 Edit a fire station

To edit a fire station, follow the steps mentioned below:

- 1. Search for a fire station by following the steps described in section 7.1.2.4.1. This displays a list of fire stations within a particular district of a state or of an entire State/UT depending on the search criteria entered.
- 2. Select the fire station for which data has to be changed by clicking the corresponding row as shown in Figure 7-11.

					CSearch Option-						
					State : Delhi		V District :	All	~	Search Show.	All
FireStationId	Reference No.	Fire Station Name	State	District	Municipality	Town/City	Latitude	Longitude	Phone No.	SurveyedBy	SurveyDate
56	DL956	Mayur Place	Delhi	East	MCD	Delhi	28.591000	77.299167		Rakesh, Ujjwal,	8/2/2011
60	DL960	Mayur Vihar	Delhi	East	MCD	Delhi	28.625167	77.307333	0112272110	Rakesh, Ujiwal,	8/2/2011
61	DL961	Tahirpur	Delhi	North East	MCD	Delhi	28.685000	77.315167	01122594700	Rakesh, Ujiwal,	8/2/2011
62	DL962	Shashtri Park	Delhi	North East	MCD	Delhi	28.675500	77.263000	01122853933	Rakesh, Ujiwal a	8/2/2011
63	Edit	Gokulpur	Delhi	North East	MCD	Delhi	28.700500	77.291333	01122815612	Rakesh, Ujiwal a	8/2/2011
64	Delete	Delhi Secretariat	Delhi	New Delhi	MCD	Delhi	28.629000	77.251833	01123392677	Rakesh, Ujjwal,	8/3/2011
65	Refresh	Shahdara	Delhi	East	MCD	Delhi	28.674167	77.284167	01122323322	Rakesh, Ujiwal,	8/2/2011
66	DL966	Geeta Colony	Delhi	East	MCD	Delhi	28.653500	77.270167	01122513330	Rakesh, Ujiwal A	8/2/2011
67	DL967	Rohini Sector 5	Delhi	North West	MCD	Delhi	28.718500	77.109833	01127042777	Anand, Abhinav	8/5/2011
68	DL968	Okhla Phase III	Delhi	South	MCD	Delhi	28.552167	77.269500	01126847511	Abhinav, Anand,	8/9/2011
69	DL969	Wazirpur	Delhi	North West	MCD	Delhi	28.701167	77.164833	01127372621	Anand and Abhin	8/5/2011
70	DL970	Keshavpuram	Delhi	West	MCD	Delhi	28.681333	77.147500	01127189090	Anand, Abhinav	8/2/2011
71	DL971	Najafgarh	Delhi	South West	MCD	Delhi	28.612000	76.976667	01128012214	Abhinav, Anand,	8/9/2011
72	DL972	Mangolpuri	Delhi	West	MCD	Delhi	28.686167	77.102000	01127015280	Anand, Abhinav	8/2/2011
73	DL973	Sarita Vihar	Delhi	South	MCD	Delhi	28.532333	77.296667	01129945066	Abhinav, Anand	8/9/2011
74	DL974	Okhla Phase I	Delhi	South	MCD	Delhi	28.523500	77.282667	01126371344	Abhinav, Anand	8/9/2011
75	DL975	Dwarka Sector 6	Delhi	South West	MCD	Delhi	28.597333	77.064167	01128082565	Anand, Ujjwal	8/10/2011
76	DL976	Narela	Delhi	North West	MCD	Delhi	28.848833	77.087833	01127785311	Abhinav, Anand	8/4/2011
78	DL978	Rashtrapati Bha	Delhi	Central	NDMC	Delhi	28.618833	77.197333	01123015321	Sushil Gupta and	9/14/2011
79	DL979	Harinagar	Delhi	West	MCD	Delhi	28.622333	77.110333	01125141433	Anand, Ujjwal	8/10/2011
80	DL980	Mathura Road	Delhi	South	MCD	Delhi	28.572500	77.257667	01126341301	Abhinav, Ujjwal a	8/9/2011
81	DL981	Janakpuri	Delhi	West	MCD	Delhi	28.621333	77.091000	01125522000	Anand, Ujjwal	8/10/2011
82	DL982	Nehru Place	Delhi	South	MCD	Delhi	28.549000	77.256333	01126445230	Abhinav, Anand	8/9/2011
	DL983	Jahangir Puri	Delhi	North West	MCD	Delhi	28.733833	77.174500	01127636006	Abhinav, Anand,	8/4/2011
83		Bhikaji Cama Place	Delhi	South West	MCD	Delhi	28.568500	77.183833	01126173583	Abhinav, Anand	8/9/2011
	DL984	erintap earna riaee									
183 184 185	DL984 DL985	Rohini Sector 16	Delhi	North West	MCD	Delhi	28.731000	77.121000	01127571194	Anand, Ujjwal, B	8/5/2011

Figure 7-11: Window showing selection of a row corresponding to a fire station having id 962 with context menu (highlighted by red rectangle)

- 3. Right click on any of the columns of the selected row to display the Context menu as shown in Figure 7-11 (red rectangle highlight).
- 4. Click on the Edit menu to display the Fire Station Information form as shown in Figure 7-12. This form comprises of eight pages (also described under section 7.1.2.1.1).

FDSS



Fire Decision Support System

.	District : North East Municipality : MCD General Information L Name : Shashtri Park Address : Near Jag Parvez chand hospital Shashtri Park Officer in-charge Details Name : Urban						
					FS R	ef #: DL9	62
Ť		Delbi	~	City/Town :	Delhi		
				City/100011			<u> </u>
					Degree	Minute	Second
	Municipality :	MCD		Lat :	28	40.53	0.0
				Lon :	77	15.78	0.0
1	General Inform	ation					
	Name :	Shashtri Pa	ark				
	Address :	Near Jag F Shashtri Pa	Parvez chand hospital ark				
	Phone No. : 1)		01122853933		2)		
	Fax No. :			Emergency N	o. :		101
	FS Type :	Urban	~				
Y	Officer in-charg	je Details					1
	Name :	Udai Vir Sir	ngh Tomar				
	Designation :	S.0	*	Mobile No	D. :		9868154496
	Administrative I	Details	-				
	Admin Depa	rtment :	State Goverment	*			
	Officer	Name :	G.C. Mishra				
	Desig	nation :	C.F.0	*			
	Mobi	le No. :	9868022790				
	Address/lo	cation :	Delhi Fire Service Head	d Quarters		1	
	No. of fire s	tation falling) under above administrat	ion/ jurisdiction	: 52	2	
	Surveyed By :	Rakesh, l	Jijwal and Abhinav	S	urvey Date :	8/ 2/201	1 💌
					Next	>>	Cancel
F	Page 1 of 8						
	-3						

Figure 7-12: Form showing general information of Shashtri Park fire station, Delhi

5. Go to the page where information needs to be edited.¹² Make the required changes. As an example, the phone number can be edited by making the required changes in the Phone No.: 1 field as highlighted by the red rectangle in Figure 7-12.

¹² For reference see the title at every page of form.



- 6. To save the changes made, the user needs to navigate to the last page by clicking the Next button given on each page.
- 7. Click the Update button on the last page as shown in Figure 7-13 (highlighted by red rectangle) to make the changes applicable.

	Month	Total Calls(A+B+C+D)	Total Fire Incidence calls(A)	Residential	Industrial	Institutional/comme	Others(fire Incidence)	Total Rescue Incidence(B)	Road Accidents	Building collapse
	Jan 2008	2	2	0	1	0	1	0	0	0
	Dec 2009	36	17	4	2	1	10	19	0	0
	Mar 2011	17	17	14	1	0	2	0	0	0
	May 2011	15	15	11	3	0	1	0	0	0
	Nov 2010	36	12	5	1	0	6	24	0	0
	Dec 2010	40	16	4	5	0	7	24	0	0
	May 2008	9	3	1	1	0	1	6	0	0
	.lan 2011	19	16	7	2	n	7	3	0	0
-	Mear Year	Total no. of small Fire	Total no. of Medium Fire	Total No. of Serious fire	Brief descriptio	n of Major Fire Incidence				
-		Total no. of			Brief descriptio	n of Major Fire Incidence				
-		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				
		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				
-		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				
		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				
		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				
		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				
-	Month Data (A,B + C+D) Incidence calls(A) Residential Industrial Institutional/comme Offension Toda (A) Data (A)									
		Total no. of small Fire	Medium Fire	Serious fire	Brief descriptio	n of Major Fire Incidence				

Figure 7-13: Form showing Fire Calls and Incidence Details page with update button (highlighted by red rectangle)

8. A message is displayed as shown in Figure 7-14. Click the OK button to save the changes made.

	×
ation Updated succes	sfully
ОК	
	ation Updated succes

Figure 7-14: Message box showing message to update fire station details

7.1.2.4.2.2 Delete a fire station

To delete a fire station, follow the steps mentioned below:

- 1. Repeat steps 1 to 3 of section 7.1.2.4.2.1
- 2. Click on Delete in the Context menu. A dialog box, as shown in Figure 7-15, is displayed.





Figure 7-15: Dialog box prompting user to confirm or ignore deletion of selected fire station

3. Click the Yes button to delete a fire station from the table. Click the No button to ignore deletion command.

Note: The Context menu also has a Refresh option. Clicking this option reloads the last saved data and displays it in the Data frame.

7.2 Master Table Editing Tool

The Master Table editing tool provides users with the capability to edit the data in the database. This database comprises of two master tables, namely, Ideal Fire stations and New Proposed Fire Stations that contain results of response time analysis and ideal count of firefighting vehicles, equipment, manpower etc.

7.2.1 Opening the tool

Double click on the Master Table editing tool icon **used** to start the application. The Master Table Editing Tool window is displayed as shown in Figure 7-16.

Master Table Editing Tool		
Master Data Table :	Search Option State : Andaman and Nicobar Islands V District : All	Search Show All
Fire Station Details		
	Add/Edit	Delete Save Close

Figure 7-16: Master Table editing tool UI



7.2.2 Master Table Editing Tool interface

The interface is divided into 3 parts as shown in Figure 7-17.

- 1. Master Data Table frame
- 2. Search Option frame
- 3. Fire Station Details frame

7.2.2.1 Master Data Table frame

Master data table comprises of two tables viz., Ideal Fire Stations and New Proposed Fire Stations. To view a list of ideal fire stations, select the Ideal Fire Station option from the Master Data Table dropdown as shown in Figure 7-17 (highlighted by red rectangle). It generates a list of all the ideal fire stations in India and displays the list in the Fire Station Details frame (highlighted by the green rectangle in Figure 7-17).

aster Data Ta e Station Deta	Select Ideal Fire Satio	ns	Search Option State : Andam	nan and Nicobar Islands	District :	All	•	Search Shov	v All
District	DistrictC		FSRefNo	Name	TOWNCITY	Urban/Rural	Estimated Total Pumping units	Estimated Total Pumping units with reserve	
Adilabad	595	1501	AP1501	Echoda Fire Stati	0	Rural	1.9	2	
Adilabad	595	1458	AP1458	Mancherial Fire S	0	Urban	2.9	3	
Adilabad	595	1503	AP1503	Jannaram Fire St	0	Rural	2.7	3	2
Adilabad	595	1463	AP1463	Bellampally Fire S	0	Urban	1.6	2	1
Adilabad	595	1504	AP1504	Utnoor Fire Station	0	Rural	2	2	
Adilabad	595	1465	AP1465	Sirpur khagaznag	0	Urban	1.1	1	
Adilabad	595	1456	AP1464	Asifabad Fire Stat	0	Rural	2.3	3	
Agra	596	1332	UP1332	Shastripuram Fire	0	Urban	0.6	1	
Agra	596	1334	UP1334	Bah Fire Station	0	Urban	1.1	1	
Agra	596	1338	UP1338	Temporary Fire St	0	Urban	3.8	5	1
Agra	596	1326	UP1326	Idgah Fire Station	0	Urban	6.9	8	1
Agra	596	1336	UP1336	Barham (Aitmadp	0	Urban	0.8	1	
Ahmada	oad 597	3747	GJ3747	Memnagar Fire St	Ahmadabad	Operational Urban	5.5	7	
Ahmada	oad 597	3745	GJ3745	Sabarmati Fire St	Ahmadabad	Operational Urban	3	4	
Ahmada	bad 597	3752	GJ3752	Jasdoha Nagar Fi	Ahmadabad	Operational Urban	6	7	
Ahmada	oad 597	3758	GJ3758	Sahpur Fire Station	Ahmadabad	Operational Urban	4	5	
Ahmada	bad 597	3759	GJ3759	Danpith Fire Stati	Ahmadabad	Operational Urban	3.3	4	

Figure 7-17: Window showing list of all ideal fire stations in India

7.2.2.2 Search Option frame

Search Option frame appears on the top right corner of the window (highlighted by the red rectangle in Figure 7-16). It allows the user to search for ideal or new proposed fire stations based on the state and district selected.

Master Table editing tool allows the users to search for fire stations in the following way:

- 1. Select Ideal Fire Stations or New Proposed Fire Stations option from Master Data Table dropdown. As an example, Ideal Fire Station option is selected (highlighted by the red rectangle in Figure 7-18).
- 2. Select a state from the State dropdown. As an example, Maharashtra state is selected (highlighted by the green rectangle in Figure 7-18).



- 3. Select a district from the District dropdown. In this dropdown the user has two options:
 - a) Select All option from the District dropdown: This option generates a list of all the fire stations within a selected State/UT.
 - **b)** Select a particular district from the District dropdown: This option generates a list of all the fire stations within a district selected, of a particular State/UT. As an example, district Pune is selected (highlighted by the blue rectangle in Figure 7-18).
- 4. Click the Search button as shown in Figure 7-18 (highlighted by the pink rectangle) to generate a list of fire stations based on the state and district selected in steps 1 and 2 above.
- 5. To view a list of all the ideal or new proposed fire stations of India, click on the Show All button in the Search Option frame as shown in Figure 7-18 (highlighted by the yellow rectangle).

	L	deal Fire Sations	~	State : Mahar	ashtra	District Pune		Search	Show All
ire S	Station Details	DistrictCode	FireStationId	FSRefNo	Name	TOWNCITY	Urban/Rural	Estimated Total Pumping units	Estimat Total P units w reserve
	Pune	1029	470	MH470	Erandawana Fire	Pune (CB)	Urban		
	Pune	1029	611	MH611	MIDC Fire Station	Daund (M Cl)	Urban		
	Pune	1029	443	MH443	Pradhikaran Fire	Pimpri Chinchwad (M	Urban		
	Pune	1029	465	MH465	Singhad Road Fir	Pune (M Corp.)	Urban		
	Pune	1029	460	MH460	Aundh Fire Station	Pune (M Corp.)	Urban		
	Pune	1029	430	MH430	Pimpri Main Fire	Pimpri Chinchwad (M	Urban		
	Pune	1029	461	MH461	Kasbapeth Fire S	Pune (CB)	Urban		
	Pune	1029	671	MH671	Central Fire Station	Pune (CB)	Urban		
	Pune	1029	609	MH609	MIDC Fire Station	Pune (M Corp.)	Urban		
	Pune	1029	449	MH449	Lt. Dayaram Rajg	Pune (CB)	Urban		
	Pune	1029	444	MH444	Hadaspar Fire St	Pune (CB)	Urban		
	Pune	1029	467	MH467	Kothrud Fire Stati	Pune (CB)	Urban		
	Pune	1029	477	MH477	Baramati Nagar P	Baramati (M Cl)	Urban		
	Pune	1029	607	MH607	MIDC Fire Station	Pimpri Chinchwad (M	Urban		
	Pune	1029	458	MH458	Yerwada Fire Sta	Kirkee (CB)	Urban		
	Pune	1029	435	MH435	Bhosari Fire Station	Pimpri Chinchwad (M	Urban		
	Pune	1029	442	MH442	Rahatini Fire Stati	Pimpri Chinchwad (M	Urban		>

Figure 7-18: Window showing details of ideal fire stations in Pune, Maharashtra

7.2.2.3 Fire Station Details frame

This frame in the window lists fire station details of ideal fire stations or new proposed fire stations depending on the option selected from the Master Data Table dropdown. The user has the option to view fire station details for a particular state or district within a state with the help of the search feature discussed above in section 7.2.2.2.



7.2.2.3.1 Modifying fire station details

7.2.2.3.1.1 Adding or editing fire station details

To edit (modify) any details in the two master tables follow the steps mentioned below:

- 1. Select the table for which data needs to be edited from Master Data Table dropdown. As an example, Ideal Fire Station table has been selected as shown in Figure 7-19.
- 2. Select the state and district in which the fire station that needs to be edit exists.
- 3. Click the Search button to populate a list of all the fire stations in the Fire Station Details frame falling under the selected state and district.
- 4. Select the fire station for which details have to be changed by clicking on the corresponding row as shown in Figure 7-19. In the figure fire station with id 613 is selected.

/laste	er Data Table :	Ideal Fire Sations	~	State : Mahar	ashtra	District : Th	ane	9	Show	v All
ire St	ation Details									
	District	DistrictCode	FireStationId	FSRefNo	Name	TOWNCITY	Urban/Rural	Estimated Total Pumping units	Estimated Total Pumping units with reserve	Ţ
4	Thane	1127	613	MH613	Mira Bhayandar F	Navghar-Manikp	Urban			
Γ	Thane	1127	675	MH675	Balkum Fire station	Thane (M Corp.)	Urban			1
	Thane	1127	586	MH586	Bhiwandi Fire Sta	Khoni (CT)	Urban			1
	Thane	1127	591	MH591	Ulhasnagar Fire	Kalyan-Dombivli (Urban			3
	Thane	1127	620	MH620	Vashi Sector 16	Navi Mumbai (M	Urban			4
	Thane	1127	977	MH977	Dombivali West F	Kalyan-Dombivli (Urban			4
	Thane	1127	580	MH580	Nizampura Fire St	Shelar (CT)	Urban			2
	Thane	1127	627	MH627	Airoli Fire Station	Navi Mumbai (M	Urban			3
	Thane	1127	631	MH631	CBD Belapur Fire	Navi Mumbai (M	Urban			3
	Thane	1127	683	MH683	Kopari Fire station	Thane (M Corp.)	Urban			2
	Thane	1127	679	MH679	Jawahar Bagh Fir	Thane (M Corp.)	Urban			2
	Thane	1127	681	MH681	Mumbra Fire Stati	Navi Mumbai (M	Urban			2
	Thane	1127	599	MH599	Silvar Park Fire S	Mira-Bhayandar (Urban			2
	Thane	1127	596	MH596	Ambarnath Fire S	Ulhasnagar (M C	Urban			3
	Thane	1127	418	MH418	Adharwadi Fire &	Bhiwandi (M Cl)	Urban			1
	Thane	1127	592	MH592	Dhamankarnaka	Khoni (CT)	Urban			3
	Thane	1127	420	MH420	MIDC Fire Station	Kalyan-Dombivli (Urban			4
1	1111							2		>

Figure 7-19: Window showing selection of a row corresponding to fire station id 613

- 5. Click the Add/Edit button (blue rectangle highlight in Figure 7-19).
- 6. Double click on any of the columns in the selected row to add or edit the data. As an example, Urban/Rural column is selected (see Figure 7-20).



	L	deal Fire Sations	~	State : Mahar	ashtra	District : Th	ane	×	Search Shov	v All
re SI	ation Details	DistrictCode	FireStationId	FSRefNo	Name	TOWNCITY	Urban/Rural	Estimated Total Pumping units	Estimated Total Pumping units with reserve	1
	Thane	1127	613	MH613	Mira Bhayandar F	Navghar-Manikp	Urban			5
	Thane	1127	675	MH675	Balkum Fire station	Thane (M Corp.)	Urban			1
	Thane	1127	586	MH586	Bhiwandi Fire Sta	Khoni (CT)	Urban			1
	Thane	1127	591	MH591	Ulhasnagar Fire	Kalyan-Dombivli (Urban			3
	Thane	1127	620	MH620	Vashi Sector 16	Navi Mumbai (M	Urban			4
	Thane	1127	977	MH977	Dombivali West F	Kalyan-Dombivli (Urban			4
	Thane	1127	580	MH580	Nizampura Fire St	Shelar (CT)	Urban			2
	Thane	1127	627	MH627	Airoli Fire Station	Navi Mumbai (M	Urban			3
	Thane	1127	631	MH631	CBD Belapur Fire	Navi Mumbai (M	Urban			3
	Thane	1127	683	MH683	Kopari Fire station	Thane (M Corp.)	Urban			2
	Thane	1127	679	MH679	Jawahar Bagh Fir	Thane (M Corp.)	Urban			2
	Thane	1127	681	MH681	Mumbra Fire Stati	Navi Mumbai (M	Urban			2
	Thane	1127	599	MH599	Silvar Park Fire S	Mira-Bhayandar (Urban			2
	Thane	1127	596	MH596	Ambarnath Fire S	Ulhasnagar (M C	Urban			3
	Thane	1127	418	MH418	Adharwadi Fire &	Bhiwandi (M Cl)	Urban			1
	Thane	1127	592	MH592	Dhamankarnaka	Khoni (CT)	Urban			3
	Thane	1127	420	MH420	MIDC Fire Station	Kalyan-Dombivli (Urban			

Figure 7-20: Window showing selection of Urban/Rural column to be edited

7. Edit the details in the selected column. In the example in Figure 7-21, the selected Ideal fire station has been changed from urban to rural.



		Ideal Fire Sations	~	State : Mahara	ashtra	District : Th	ane	✓	Search Show	v Al
	on Details District	DistrictCode	FireStationId	FSRefNo	Name	TOWNCITY	Urban/Rural	Estimated Total Pumping units	Estimated Total Pumping units with reserve	
1	Thane	1127	613	MH613	Mira Bhayandar F	Navghar-Manikp	Rural			
T	[hane	1127	675	MH675	Balkum Fire station	Thane (M Corp.)	Urban			
I	Thane	1127	586	MH586	Bhiwandi Fire Sta	Khoni (CT)	Urban			
T	Thane	1127	591	MH591	Ulhasnagar Fire	Kalyan-Dombivli (Urban			
T	Thane	1127	620		ctor 16	Navi Mumbai (M	Urban			
T	Thane	1127	977		i West F	Kalyan-Dombivli (Urban			
T	Thane	1127	580	Fire station updated s	a Fire St	Shelar (CT)	Urban			
T	[hane	1127	627	ОК	Station	Navi Mumbai (M	Urban			
T	[hane	1127	631		apur Fire	Navi Mumbai (M	Urban			
T	[hane	1127	683	MH683	Kopari Fire station	Thane (M Corp.)	Urban			
T	[hane	1127	679	MH679	Jawahar Bagh Fir	Thane (M Corp.)	Urban			
T	[hane	1127	681	MH681	Mumbra Fire Stati	Navi Mumbai (M	Urban			
T	[hane	1127	599	MH599	Silvar Park Fire S	Mira-Bhayandar (Urban			
T	Thane	1127	596	MH596	Ambarnath Fire S	Ulhasnagar (M C	Urban			
T	[hane	1127	418	MH418	Adharwadi Fire &	Bhiwandi (M Cl)	Urban			
T	Thane	1127	592	MH592	Dhamankarnaka	Khoni (CT)	Urban			
T	[hane	1127	420	MH420	MIDC Fire Station	Kalyan-Dombivli (Urban			
	Ш			55 - 560 - 660 / 777						>

Figure 7-21: Window showing Urban/Rural column changed from Urban to Rural type for fire station having Id 613

- 8. Click on the Save button (as highlighted by the yellow rectangle in Figure 7-21). A message box stating that the ideal fire station details have been updated is displayed as shown by the green rectangle in Figure 7-21.
- 9. Click the OK button to update the ideal fire station details in the database.

7.2.2.3.1.2 Deleting a fire station

To delete an ideal or new proposed fire station from the database, follow the steps mentioned below:

- 1. Repeat steps 1 to 3 from section 7.2.2.3.1.1.
- 2. Select the fire station for which details have to be changed by clicking on the corresponding row as shown in Figure 7-22.
- 3. Click the Delete button as shown by the red rectangle highlight in Figure 7-22.



Aaster Data Table : Ideal Fire Sations			State : Maharashtra		District : Thane		Search Show All		
e Station Deta	ls							Estimated	
District	DistrictCode	FireStationId	FSRefNo	Name	TOWNCITY	Urban/Rural	Estimated Total Pumping units	Estimated Total Pumping units with reserve	
Thane	1127	613	MH613	Mira Bhayandar F	Navghar-Manikp	Urban			1
Thane	1127	675	MH675	Balkum Fire station	Thane (M Corp.)	Urban			
Thane	1127	586	MH586	Bhiwandi Fire Sta	Khoni (CT)	Urban			
Thane	1127	591	MH591	Ulhasnagar Fire	Kalyan-Dombivli (Urban			
Thane	1127	620	MH620	Vashi Sector 16	Navi Mumbai (M	Urban			
Thane	1127	977	MH977	Dombivali West F	Kalyan-Dombivli (Urban			
Thane	1127	580	MH580	Nizampura Fire St	Shelar (CT)	Urban			
Thane	1127	627	MH627	Airoli Fire Station	Navi Mumbai (M	Urban			
Thane	1127	631	MH631	CBD Belapur Fire	Navi Mumbai (M	Urban			
Thane	1127	683	MH683	Kopari Fire station	Thane (M Corp.)	Urban			
Thane	1127	679	MH679	Jawahar Bagh Fir	Thane (M Corp.)	Urban			
Thane	1127	681	MH681	Mumbra Fire Stati	Navi Mumbai (M	Urban			
Thane	1127	599	MH599	Silvar Park Fire S	Mira-Bhayandar (Urban			
Thane	1127	596	MH596	Ambarnath Fire S	Ulhasnagar (M C	Urban			
Thane	1127	418	MH418	Adharwadi Fire &	Bhiwandi (M Cl)	Urban			-
Thane	1127	592	MH592	Dhamankarnaka	Khoni (CT)	Urban			
Thane	1127	420	MH420	MIDC Fire Station	Kalvan-Dombivli (Urban			

Figure 7-22: Window showing selection of a row corresponding to fire station id 591

4. A dialog box, as shown in Figure 7-23, is displayed.

Delete 🛛 🛛
Do you want to delete this fire station details ?
Yes No

Figure 7-23: Dialog box displaying question to delete a fire station

- 5. Click the Yes button to delete the fire station from the table. Click the No button to ignore the deletion command.
- 6. Click the Save button, as shown in Figure 7-22, to apply the changes to the database table.







RMSI is a professional services company providing geospatial solutions and application software services to clients worldwide





Property Rights Notice: All Rights Reserved. This material contains the valuable properties and trade secrets of RMSI, embodying substantial creative efforts and confidential information, ideas and expressions, no part of which may be reproduced or transmitted in any form or by any means, electronic, mechanical or otherwise, including photocopying and recording, or in connection with any information storage or retrieval system without permission in writing from the authorized officer of RMSI, or an RMSI subsidiary.